

Citation: Roland BP, Zeccola AM, Larsen SB, Amrich CG, Talsma AD, Stuchul KA, et al. (2016) Structural and Genetic Studies Demonstrate Neurologic Dysfunction in Triosephosphate Isomerase Deficiency Is Associated with Impaired Synaptic Vesicle Dynamics. PLoS Genet 12(3): e1005941. doi:10.1371/journal.pgen.1005941

Editor: Leo J Pallanck, University of Washington, UNITED STATES

Received: September 1, 2015

Accepted: February 24, 2016

Published: March 31, 2016

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative Commons CC0</u> public domain dedication.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files or have been deposited (Protein Databank under accession code 4ZVJ).

Funding: This work was supported by a fellowship from Achievement Rewards for College Scientists: Pittsburgh Chapter [BPR]; DOE/BER grant FWP BO-70 (AH); and the National Institutes of Health [grant numbers R01 GM103369 & R01 GM108073 (MJP), R01 GM097204 (APV), R01 NS32385 (ESL), 8P41GM103473-16 (AH), and T32 GM8424-17 RESEARCH ARTICLE

Structural and Genetic Studies Demonstrate Neurologic Dysfunction in Triosephosphate Isomerase Deficiency Is Associated with Impaired Synaptic Vesicle Dynamics

Bartholomew P. Roland^{1,2^a}, Alison M. Zeccola^{1,2}, Samantha B. Larsen^{1,2}, Christopher G. Amrich³, Aaron D. Talsma^{1,2}, Kimberly A. Stuchul^{1,2}, Annie Heroux⁴, Edwin S. Levitan¹, Andrew P. VanDemark³, Michael J. Palladino^{1,2}*

 Department of Pharmacology & Chemical Biology, University of Pittsburgh School of Medicine, Pittsburgh, Pennsylvania, United States of America, 2 The Pittsburgh Institute for Neurodegenerative Diseases (PIND), University of Pittsburgh School of Medicine, Pittsburgh, Pennsylvania, United States of America,
Department of Biological Sciences, University of Pittsburgh, Pittsburgh, Pennsylvania, United States of America, 4 Energy Sciences Directorate/Photon Science Division, Brookhaven National Laboratory, Upton, New York, United States of America

¤ Current Address: Department of Biological Sciences, Vanderbilt University, Nashville, Tennessee, United States of America

* mjp44@pitt.edu

Abstract

Triosephosphate isomerase (TPI) deficiency is a poorly understood disease characterized by hemolytic anemia, cardiomyopathy, neurologic dysfunction, and early death. TPI deficiency is one of a group of diseases known as glycolytic enzymopathies, but is unique for its severe patient neuropathology and early mortality. The disease is caused by missense mutations and dysfunction in the glycolytic enzyme, TPI. Previous studies have detailed structural and catalytic changes elicited by disease-associated TPI substitutions, and samples of patient erythrocytes have yielded insight into patient hemolytic anemia; however, the neuropathophysiology of this disease remains a mystery. This study combines structural, biochemical, and genetic approaches to demonstrate that perturbations of the TPI dimer interface are sufficient to elicit TPI deficiency neuropathogenesis. The present study demonstrates that neurologic dysfunction resulting from TPI deficiency is characterized by synaptic vesicle dysfunction, and can be attenuated with catalytically inactive TPI. Collectively, our findings are the first to identify, to our knowledge, a functional synaptic defect in TPI deficiency derived from molecular changes in the TPI dimer interface.

Author Summary

Glycolysis is the metabolic pathway that cells use to break down the sugar glucose, and mutations in the genes that control the glycolytic pathway elicit a collection of diseases known as glycolytic enzymopathies. Glycolytic enzymopathies are rare genetic diseases



(BPR)]. Data was collected at beamline X25 of the National Synchrotron Light Source (NSLS), which was supported under Contract No. DE-AC02-98CH10886 of U.S. Department of Energy Office of Basic Energy Sciences. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

that lead to the degeneration of patient red blood cells. *Triosephosphate isomerase* is a gene that encodes a part of this glycolytic process, and patients with mutations in this gene experience the typical blood disorder, as well as severe neurologic dysfunction and often infant death. Until now, a molecular source of neurologic dysfunction in *triosephosphate isomerase* mutants was unknown. We have discovered that mutations that disrupt the self-association of the triosephosphate isomerase enzyme lead to neurologic dysfunction in fruit flies. This neurologic dysfunction is characterized by the abnormal cycling of neuronal vesicles that contain neurotransmitters. Given the evolutionary conservation of *triosephosphate isomerase* and neuronal synaptic function, we believe that these observations represent the dysfunction seen in human patients.

Introduction

Triosephosphate isomerase (TPI) is a glycolytic enzyme that converts dihydroxyacetone phosphate (DHAP) into glyceraldehyde-3 phosphate (GAP). TPI is a non-linear member of the glycolytic pathway, enhancing the efficiency of the catabolic process, and several missense mutations within *TPI* lead to a disease known as TPI deficiency [1]. TPI deficiency is one member of a group of disorders caused by mutations in glycolytic enzymes, collectively called glycolytic enzymopathies. Glycolytic enzymopathies are largely characterized as blood disorders, with all patients experiencing hemolytic anemia [2,3]. TPI deficiency is one of the few glycolytic diseases associated with patient neurologic dysfunction, and by far the most severe [1,4]. Clinical examinations of TPI deficiency patients have established that this disease is often characterized by episodic seizures, periodic dystonia, and progressive weakness and flaccidity in extremities [5–11]. Cellular studies have centered on erythrocytes and lymphocytes, leaving it unclear how TPI molecular dysfunction influences the nervous system. Further, the absence of neurologic dysfunction in many other glycolytic enzymopathies has made it unclear whether these symptoms are related to glycolysis or an as-yet-unidentified function of TPI.

Several reports have suggested that TPI deficiency is a disease caused by changes in protein conformation rather than metabolic defects [4,12–14]. A human protein structure and yeast genetic studies have asserted that defects in TPI dimerization are the primary determinants of pathology [15,16], revealing no catalytic defects *in vitro* or from cell lysate. However, not all TPI deficiency mutations lack catalytic defects. An erythrocyte study examining two Hungarian brothers with identical *TPI* alleles revealed equivalent reductions in TPI activity in both individuals [17], yet one exhibited severe neurologic dysfunction and the other was asymptomatic. Further, a recent structural study demonstrated that the hTPI^{I170V} substitution significantly altered enzyme kinetics and protein stability through a molecular alteration near the catalytic pocket [18]. Collectively, each of these studies failed to establish a causal relationship between TPI activity and disease.

Previously, a pathogenic substitution in *Drosophila TPI* ($dTPI^{M80T}$) was identified, eliciting mechanical- and thermal-stress dependent paralysis [19,20]. These behavioral phenotypes have been independently established as hallmarks of neurologic dysfunction and each has been used in forward genetic screens to identify novel components of neuronal transmission [21–23]. The $dTPI^{M80T}$ allele was identified in such a screen [20,24], and to-date *D. melanogaster* is the only model organism to exhibit neurologic dysfunction caused by TPI deficiency. The $dTPI^{M80T}$ protein was found to be prematurely degraded with reduced catalytic activity [25,26]. This reduction in catalytic activity was shown to inhibit glycolytic flux as well as induce metabolic stress [19,25], yet did not change ATP levels *in vivo* [20]. Subsequent studies

demonstrated that the *dTPI^{M80T}* point mutation could be complemented by the addition of a catalytically inactive *TPI* without increasing lysate isomerase activity or alleviating metabolic stress [25], suggesting that dTPI^{M80T} may elicit pathology through a change in protein conformation. Thus, we initially examined the molecular source of *dTPI^{M80T}* pathology.

To determine whether *Drosophila* TPI deficiency was caused by changes in protein conformation, we purified and assessed the physical characteristics of hTPI^{M82T}, the human equivalent of dTPI^{M80T}, and revealed impaired TPI dimerization. These results were further supported when independent alleles bearing mutations at the TPI dimer interface phenocopied the *Drosophila* behavioral dysfunction seen in the *dTPI^{M80T}* allele. These experiments provided novel insight into the pathogenesis of TPI deficiency leading to the conclusion that alterations of TPI dimerization are sufficient to elicit neuropathology.

Defining the molecular source of TPI neurologic dysfunction led to the generation of new alleles containing dimer-interface mutations. These novel *TPI* alleles were characterized by extreme behavioral defects, directing new investigations into the neuropathogenic mechanism of TPI deficiency. An examination of vesicle dynamics at the larval neuromuscular junction (NMJ) revealed a severe impairment that appears to be related to vesicle recycling. Further, complementation with a *TPI* allele encoding catalytically inactive TPI rescued both synaptic dysfunction and behavior, thereby characterizing a cellular mechanism of TPI deficiency neuropathology.

Collectively, the results of this study support the conclusion that an improperly formed dimer interface is sufficient to elicit TPI deficiency neuropathology. Further, our experiments establish that a functional synaptic defect occurs in our *Drosophila* model of TPI deficiency.

Results

hTPI^{M82T} impairs TPI homodimerization

TPI is a homodimeric enzyme with catalytic sites in the C-terminal of a triose isomerase (TIM) barrel tertiary structural motif [27]. Each catalytic site is rigidified through dimerization to increase catalytic turnover, yet each active site works independently [28–30]. A $dTPI^{M80T}$ substitution was previously isolated and demonstrated to elicit pathology in a *Drosophila* model of TPI deficiency [19,20]. The $dTPI^{M80T}$ substitution is physically located in a solvent-exposed region of the protein near the dimer interface [25]. Numerous misfolding events could be hypothesized to occur as a function of the TPI^{M80T} substitution, among them alterations of dimerization [15,16] and aggregation [31]. To examine the structural change elicited by M80T *in vitro* we purified *Drosophila* dTPI^{M80T}.

Previous purification experiments had yielded *Drosophila* TPI enzyme, but these samples proved aggregation prone at high concentrations. Conversely, purified human TPI (hTPI) was well-behaved; therefore, in order to physically characterize TPI we studied the human protein *in vitro*. To validate the use of human protein *in vivo* we generated human WT ($hTPI^{WT}$) and human M80T ($hTPI^{M82T}$) alleles in the *Drosophila TPI* gene locus using an established genomic engineering (GE) system [25]; the hTPI^{M82T} substitution is equivalent to dTPI^{M80T} (Fig 1A). We found that $hTPI^{M82T}$ was able to recapitulate the disease phenotypes observed in $dTPI^{M80T}$ (S1 Fig). The phenotypes of $hTPI^{M82T}$ were remarkably similar but less severe than $dTPI^{M80T}$, possibly due to subtle organism-specific changes in the dimer interface [32]. Confirmation that $hTPI^{M82T}$ pathologically phenocopied $dTPI^{M80T}$ indicated that any conformational change elicited by dTPI^{M80T} was likely retained in the human protein.

We utilized dynamic light scattering (DLS) to examine potential conformational differences between $hTPI^{WT}$ and $hTPI^{M82T}$. Analyses of 15 μ M solutions of $hTPI^{WT}$ revealed a hydrodynamic radius of 4.3 ± 0.08 nm, while $hTPI^{M82T}$ exhibited a significant reduction to 3.3 ± 0.06 nm



Fig 1. hTPI^{M82T} elicits a conformational change in TPI resulting in reduced dimerization. (A) Sequence alignment of *D.melanogaster* and *H.sapiens* TPI protein sequence with asterisks highlighting residues of interest. (B) The hTPI^{M82T} mutation confers a reduction in mean protein hydrodynamic radius as measured by dynamic light scattering. (C) Intensity correlation plots reveal a largely monodisperse hTPI^{WT} population and polydisperse hTPI^{M82T} population. (D) Gel filtration indicates a change in monomer:dimer ratios elicited by hTPI^{M82T} with relative quantification (inset). n \geq 3, comparisons were made using Student's T test, *** indicates p<0.001.

(Fig 1B); these results were consistent across two additional protein concentrations, $3.75 \,\mu$ M and $30 \,\mu$ M (Fig 1B). The linear slope generated by plotting the intensity correlation data suggested the hTPI^{WT} sample was largely monodispersed, much like that of the 15 μ M sample of bovine serum albumin (Fig 1C). Conversely, hTPI^{M82T} samples exhibited a non-linear slope (Fig 1C), suggesting the possibility of a polydisperse protein population. Polydisperse protein populations indicated the sample was a mixed population in solution, and the observed reduction in TPI mean hydrodynamic radius suggested the sample could be a mixture of monomer and dimer TPI species.

We examined enzyme dimerization by assessing protein size via gel filtration chromatography. A standard curve was used to establish column resolution, 15 μ M samples of hTPI^{WT} and hTPI^{M82T} were injected onto the gel filtration column, and their migration monitored by UV light at 280 nm. hTPI^{WT} samples separated into two distinct peaks–one at ~24 min. and another at ~27 min. corresponding to ~50 kDa and ~28 kDa, respectively (Fig 1D). Dimeric and monomeric hTPI are 54 and 27 kDa, respectively. Integrating the peak areas revealed an ~80:20 split in dimer:monomer ratio of hTPI^{WT} (Fig 1D) inset). In contrast, the majority of the hTPI^{M82T} sample eluted at 27 min., resulting in a ~5:95 dimer:monomer ratio (Fig 1D) inset). These data led us to conclude that the hTPI^{M82T} substitution elicited a dramatic conformational change in TPI resulting in a disruption of dimerization. Interestingly, the gel filtration results did not precisely reflect the monodisperse vs. polydisperse observations of the DLS experiments; we believe this could be due to dilution effects as the proteins migrated over the large gel filtration column.

TPI dimer interface mutants recapitulate dTPI^{M80T} neuropathology

Having established that the hTPI^{M82T} mutation alters enzyme dimerization *in vitro*, we sought to assess whether other substitutions at the TPI dimer interface were sufficient to elicit neuropathology. Two novel *TPI* alleles ($dTPI^{T73R}$ and $dTPI^{G74E}$) were generated using GE. Our hTPI dimer analyses (S2 Fig) and data from previous TPI studies [28] indicated these substitutions would result in dimer defective TPI. $dTPI^{T73R}$ and $dTPI^{G74E}$ dimer interface mutants elicited a more severe pathology than $dTPI^{M80T}$, and stocks required maintenance over balancer chromosomes due to their poor viability. Test crosses of balanced stocks yielded significantly fewer homozygous animals than the Mendelian predicted 33%, and homozygous animals were extremely short-lived (Fig 2A), with median lifespans of 2 and 5 days for $dTPI^{T73R}$ and $dTPI^{G74E}$, respectively.

Mechanical- and thermal stress-dependent behavioral defects were assessed at Day 1 and Day 2, respectively, as these phenotypes have been demonstrated to be hallmarks of *Drosophila* TPI deficiency [19,20,25,26,33,34]. $dTPI^{M80T}$ was previously described to exhibit a modest phenotype at early time points [19], and these data corroborate our analyses of the GE $dTPI^{M80T}$ allele (Fig 2B and 2C). Comparatively, the dimer interface mutants displayed a more severe degree of behavioral dysfunction than that seen in $dTPI^{M80T}$ (Fig 2B and 2C). These data support the hypothesis that mutations at the dimer interface are sufficient to induce neurologic dysfunction.

Lysate isomerase activity was compared between samples taken from animals homozygous for the dimer interface mutants. First, it was noted that all dimer interface mutants exhibited reductions in TPI activity (Fig 2D). However, a comparison of the $dTPI^{M80T}$, $dTPI^{T73R}$, and $dTPI^{G74E}$ lysates revealed a striking observation–the least phenotypically severe mutation ($dTPI^{M80T}$) was characterized by the lowest isomerase activity (Fig 2D). These data support previous observations that TPI activity does not predict the presence or severity of TPI deficiency [25].



Fig 2. Mutations affecting the TPI dimer interface recapitulate $dTPI^{M80T}$ phenotypes. (A) $dTPI^{T73R}$ and $dTPI^{G74E}$ homozygotes display severely reduced lifespans, n>150. (B) Dimer interface mutations exhibit severe mechanical stress at Day 1 and (C) thermal stress sensitivity at Day 2, n>30. Thermal stress paralysis times at 360 sec. represent wild type behavior, the assay was stopped at 6 min. (D) Both $dTPI^{T73R}$ and $dTPI^{G74E}$ homozygotes display reduced lysate isomerase activity, n \geq 3. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, and lifespans by a Log-rank (Mantel-Cox) survival test, ** indicated p<0.01, *** p<0.001.

Many conformational diseases are elicited through changes in protein structure and stability leading to misfolding, then either sequestration and degradation, or aggregation [35]. First, we examined whether these new dimer interface alleles produced robust levels of TPI protein. We determined TPI levels in our dimer interface mutants as previously [34], and found that both $dTPI^{T73R}$ and $dTPI^{G74E}$ homozygotes exhibited reduced protein levels (Fig 3A and 3B).

It has previously been shown that TPI has the capacity to aggregate and thereby seed the aggregation of other proteins such as tau [31]. When measuring protein levels via SDS-PAGE, it is important to note that not all aggregate species are SDS soluble and a reduction in protein levels can indicate that the aggregates are not passing through the gel matrix. To determine whether the dTPI^{T73R} and dTPI^{G74E} proteins aggregate we used a dot-blot filter trap assay to assess retention differences between TPI mutant isoforms, as performed previously [36]. Lysates were collected from homozygous animals, and PC12 cell lysates expressing EGFP-huntingtin-Q97 (GFP htt-Q97) were used as a positive aggregation control. The results indicate that little TPI was trapped on the 200 nm filter, yet each sample showed a concentration-dependent increase in signal (Fig 3C). Importantly, no differences were observed in TPI signal between the WT and mutant alleles (Fig 3C). These data support similar findings established by sedimentation assays performed on $dTPI^{M80T}$ [34], and led us to conclude that although these dimer interface mutants display reduced protein levels via SDS-PAGE, this is not due to the insolubility of large aggregates.

To date, all but one study examining TPI deficiency in *Drosophila* have highlighted a reduction in TPI protein levels in disease-associated alleles [18-20,25,26]. To independently examine



Fig 3. Mutations at the TPI dimer interface reduce TPI levels *in vivo* without aggregation. (A) $dTPI^{CT3R}$ and $dTPI^{CT4R}$ homozygote animal lysates display reduced protein levels with (B) quantification normalized to WT and an ATPalpha loading control, n = 3. (C) The reduction in SDS-soluble TPI is not caused by protein aggregation; increasing amounts of lysate were loaded and show no differences in filter-trapped TPI across all genotypes, (D) 10µg of huntingtin exon1-GFP lysate displayed robust retention on the filter, n = 2. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, *** p<0.001 relative to WT.

doi:10.1371/journal.pgen.1005941.g003

the importance of TPI protein levels *in vivo*, we employed the GAL4-UAS expression system to knock down wild type (WT) TPI using a UAS-RNAi line directed toward *dTPI* messenger RNA (mRNA) [<u>37</u>]. These lines were driven with actin-GAL4 + UAS-GAL4 (actin/UAS-GAL4) to obtain a dramatic reduction of TPI in all tissues.

Using UAS-RNAi in conjunction with actin/UAS-GAL4, we found that *w;actin-GAL4*, *UAS-GAL4/+;UAS-RNAi*^{TPI}/+ animals exhibited a dramatic reduction in TPI protein levels similar to that seen in head and thorax tissue from *w;;dTPI*^{M80T} homozygotes (S3A and S3B Fig). Head and body tissues were assessed separately to ensure equivalent knockdown in both tissues. Next, we examined animal behavior in these knockdown populations to determine whether depletion of cellular TPI was sufficient to elicit TPI deficiency behavioral abnormalities. Mechanical stress responses were used to quantify behavioral dysfunction. None of the knockdown genotypes exhibited abnormal mechanical-stress dependent responses (S3C Fig). No paralysis or seizure-like activity was observed in the knockdown animals consistently dwelling near the bottom of the vial relative to their *TPI*⁺ and UAS only controls. These observations suggested that a general depletion of TPI is not sufficient to elicit paralysis or seizure-like locomotor dysfunction, yet do not exclude the possibility that changes in protein conformation, localized subcellular depletions, or changes in protein stability may play a greater role in animal pathology.

A catalytically inactive TPI enzyme suppresses TPI deficiency

Previous work demonstrated that a catalytically inactive allele of TPI ($dTPI^{\Delta cat}$) complemented the behavior and longevity defects of the $dTPI^{M80T}$ allele [25], a mutation now established to disrupt enzyme dimerization. This previous study suggested that TPI deficiency is a loss-of-function disease caused by either i) the depletion of cellular TPI, or ii) a conformational change that could be rescued through the addition of a properly folded yet catalytically open/inactive



Fig 4. A catalytically inactive allele attenuates TPI dimer mutant behavior and longevity. (A) $dTPI^{\Delta cat}$ complements $dTPI^{T73R}$ mechanical and (B) thermal stress sensitivity and (C) longevity, and (D,E) partially attenuates behavior and (F) longevity defects of $dTPI^{G74E}$. Thermal stress paralysis times at 360 sec. represent wild type behavior, the assay was stopped at 6 min. n \geq 30 for behavior, and n \geq 90 for lifespans. Behavior was compared using a One-way ANOVA with Tukey's post hoc test, and lifespans by a Log-rank (Mantel-Cox) survival test, *ns* indicates no significance, and *** p<0.001.

doi:10.1371/journal.pgen.1005941.g004

isoform [25]. Having utilized knockdown strategies to examine the necessity of total TPI levels, we sought to confirm the capacity of $dTPI^{\Delta cat}$ (Lys-to-Met, position 11, Fig 1A) to complement additional dimer-interface mutations. To evaluate whether $dTPI^{\Delta cat}$ was sufficient to support normal behavior and longevity, dTPI⁺/dTPI⁺, dTPI⁺/dTPI^{T73R}, dTPI^{T73R}, dTPI^{T7}, dTPI^{T7}, dTPI^{T7}, dTPI^{T7}, dTPI^{T7}, dTPI^{T7}, dTPI^{T7}, $dTPI^{\Delta cat}$, $dTPI^+/dTPI^{G74E}$, $dTPI^{G74E}/dTPI^{G74E}$, and $dTPI^{G74E}/dTPI^{\Delta cat}$ animals were collected and tested as outlined above. These experiments demonstrated that the *dTPI*^{T73R} allele was nearly fully complemented by $dTPI^{\Delta cat}$ (Fig 4A-4C), similar to the results found with $dTPI^{M80T}$ [25]. It should be noted that this complementation was not fully penetrant; 5 out of the 30 $dTPI^{T73R}/dTPI^{\Delta cat}$ animals did paralyze after an extended thermal stress period (Fig 4B). The penetrance of the thermal stress complementation is reflected in an increased time to paralysis relative to the homozygous mutant animals (Fig 4B and 4E). dTPI^{G74E} was also complemented by $dTPI^{\Delta cat}$, although more modestly than was observed for $dTPI^{T73R}$. Mechanical stress responses were unchanged in $dTPI^{G74E}/dTPI^{\Delta cat}$ relative to $dTPI^{G74E}$ homozygotes, though the penetrance of thermal stress sensitivity was decreased to 20 out of 30 animals, and the median lifespan of the $dTPI^{G74E}$ mutants was extended from 5 to 21 days (Fig 4D-4F). Importantly, neither of the dimer interface mutants elicited dominant negative effects within the *dTPI*⁺ heterozygotes; to the contrary, *dTPI*^{T73R} and *dTPI*^{G74E} promoted a significant increase in animal health, extending the median 48 day $dTPI^+/dTPI^+$ lifespans to 77 and 71 days, respectively (Fig 4C and 4F). In toto, dTPI^{Δcat} partially but significantly complemented each of the new TPI dimer alleles.

	hTPI ^{∆CAT}
Data collection	
Space group	P 2 ₁ 2 ₁ 2 ₁
Wavelength (Å)	
Cell dimensions	
a, b, c (Å)	65.1, 73.5, 92.8
α, β, γ (°)	90.0, 90.0, 90.0
Unique Reflections	48,935
Resolution (Å)	50.0–1.70 (1.73–1.70
R _{merge} (%)	6.3 (58.6)
Ι / σΙ	42.0 (2.9)
Completeness (%)	100 (100)
Redundancy	12.3 (11.2)
Refinement	
Resolution (Å)	48.7–1.70 (1.73–1.70
R _{work} / R _{free} (%)	15.8/19.6 (21.0/29.4)
FOM	0.87 (0.82)
Number of. atoms	
Protein	7,432
Other	16
Solvent	455
B-factors	
Protein	30.6
Other	54.1
Solvent	37.8
R.m.s. deviations	
Bond lengths (Å)	0.017
Bond angles (°)	1.25
Rotamer outliers (%)	1.81
Ramachandrian	
Outliers (%)	0.20
Allowed (%)	1.43
Favored (%)	98.4

Table 1.	Crystallographic	Data collection and	Refinement statistics
----------	------------------	---------------------	-----------------------

*Values in parentheses are for highest-resolution shell.

doi:10.1371/journal.pgen.1005941.t001

$hTPI^{\Delta cat}$ crystal structure

Our experiments with TPI dimer mutations demonstrated that alterations of the dimer interface were sufficient to elicit TPI neurologic dysfunction, and that these phenotypes were able to be complemented with $dTPI^{\Delta cat}$. To address how the $TPI^{\Delta cat}$ substitution may influence its structure, we purified, crystallized, and determined the structure of $hTPI^{\Delta cat}$ at 1.7Å resolution, refining against native data to R_{work} and R_{free} values of 15.8%, and 19.6%, respectively (<u>Table 1</u>). These crystals grew in conditions that were nearly identical to conditions in which we have previously determined the structure of wild-type human TPI [<u>18</u>], minimizing the effects that changes in the crystallization condition or crystal packing might have on the resulting structure. While the overall fold of $hTPI^{\Delta cat}$ is highly similar to wild-type (r.m.s.d of 0.35 Å over all atoms) there are a number of important differences within the catalytic pocket and neighboring regions. First, the active site pocket of our previous $hTPI^{WT}$ structure contained a



Fig 5. hTPl^{Δcat} crystal structure reveals a catalytically incompetent enzyme with an unaltered dimer interface. (A) The hTPl^{Δcat} adopts an open lid conformation. Superposition of hTPl^{WT} (grey) and hTPl^{Δcat} (blue) structures. Loop3 (tan), the position of the lid covering the TPl active site (magenta and pink), and the locations of bromide and phosphate ions from the active site pocket of hTPl^{WT} are indicated. New hydrophobic interactions that help to reposition M13 are shown (green). (B) The dimer interface of hTPl^{Δcat} is unchanged relative to hTPl^{WT}. Superposition of hTPl^{WT} (grey) and hTPl^{Δcat} shown as in (A), with monomer subunits of hTPl^{Δcat} dimer in blue and tan. Key dimer interface residues and loops are indicated.

highly ordered phosphate and bromide ion located where the phosphate and triose groups of the natural substrate, DHAP, would be located [18]. In contrast, the active site pocket of $hTPI^{\Delta-1}$ ^{cat} was filled with solvent. At the site of the hTPI^{K13M} substitution (hTPI^{Δcat}), the M13 side chain adopts a different conformation than its lysine counterpart, shifting 4 Å away from the catalytic site and interacting with N11, G233 and L236 at the back of the pocket (Fig 5A). The sidechain positions of important active site residues S96 and E165 are also altered in $hTPI^{\Delta cat}$, breaking critical solvent networks and shifting E165 2.7 Å away from the position it adopts in wild-type TPI [18] and substrate analog bound structures [38-40] (Fig 5A). Lastly, the lid moves as much as 7 Å away from the active site pocket, adopting an open conformation [40,41] (Fig 5A) and corroborating established kinetic data demonstrating that this enzyme is catalytically inactive [42]. These data are in agreement with a structure of yeast TPI^{K12M,G15A} containing two mutations within the active site [43], but were an important control to isolate the structural impact of hTPI^{K13M}. Importantly, examinations of the dimer interface of hTPI^{Δ-} ^{cat} revealed that it is unchanged relative to hTPI^{WT} (Fig 5B). The peptide backbone and side chains of Loop3 form the majority of the TPI dimer interface, and as shown previously, perturbations of this loop disrupt TPI dimer stability [28-30,44]. The new crystal structure revealed that the backbone of Loop3 along with important side chains M14, T75, G76, M82, and E104, are unaltered in hTPI^{Δcat} (Fig 5B). These structural data indicated that TPI^{Δcat} homodimers are catalytically inactive, with no observable alterations of the overall folding of the monomers or their dimeric assembly (Fig 5B).

dTPI^{∆cat} forms heterodimers with mutant TPI subunits

Complementation of the *TPI* dimer mutant alleles with $dTPI^{\Delta cat}$ suggested a physical and/or functional interaction between the two enzymes. Given the dimeric nature of TPI, we sought to first examine physical interactions between TPI species. All of the dimer-interface substitutions used in this study disrupt homodimerization (Figs <u>1</u> and <u>S2</u>), though no experiments had yet addressed how these alterations may change heterodimerization with $dTPI^{\Delta cat}$. These putative heterodimers could support or inhibit a critical function of TPI.

To examine heterodimer formation *in vivo*, we measured the capacity of the dimer-mutant TPI isoforms to co-precipitate using a C-terminal Cerulean cyan fluorescent protein (CFP) tagged variant of $dTPI^{Acat-CFP}$; this allele was previously confirmed to complement $dTPI^{M80T}$

[25]. anti-GFP was covalently conjugated to the AminoLink resin, and the CFP tag was immunoprecipitated (IPed) in $dTPI^{WT}/dTPI^{Acat-CFP}$, $dTPI^{M80T}/dTPI^{Acat-CFP}$, $dTPI^{T73R}/dTPI^{Acat-CFP}$, and $dTPI^{G74E}/dTPI^{Acat-CFP}$ animal lysates and probed. Unconjugated resin was incubated with $dTPI^{WT}/dTPI^{Acat-CFP}$ lysate and used as a negative control (-) (Fig 6A, IP). Upon elution and SDS-PAGE separation, protein size was used to discriminate between the tagged and untagged TPI isoforms; the CFP tag roughly doubled the molecular weight of dTPI-CFP monomer (~50 kDa) relative to dTPI monomer (~25kD) (Fig 6A, Input).

Robust amounts of dTPI^{WT} precipitated with dTPI^{Δcat-CFP}, establishing substantial heterodimerization between the two species (Fig 6A and 6B) in agreement with the similarities between their respective dimer interfaces (Fig 5B). Conversely, dTPI^{M80T} and dTPI^{T73R} displayed markedly reduced associations with dTPI^{Δcat-CFP}, corroborating their previously established dimerization deficiencies and reflecting their overall prevalence in the lysate (Fig 6A). Finally, it was surprising to see that dTPI^{G74E} produced heterodimerization similar to that seen in dTPI^{WT} (Fig 6A and 6B); it was predicted that the rotational flexibility of G74 was necessary for the appropriate positioning of loop 3 and establishment/rigidification of the dimer interface [28].

The coIP experiments suggested that the TPI species responsible for $dTPI^{T73R}$ phenotype suppression in the animals was not a $dTPI^{\Delta cat}$ heterodimer; the heterodimer was a very small fraction of the total TPI enzyme in lysate (Fig 6). Conversely, the $dTPI^{\Delta cat-CFP}$:: $dTPI^{G74E}$ heterodimer existed as a substantial fraction of the total TPI (Fig 6), yet exhibited modest complementation of the abnormal behavioral phenotypes (Fig 3). The substantial and unanticipated presence of the $dTPI^{\Delta cat-CFP}$:: $dTPI^{G74E}$ heterodimer could indicate an allele-specific dominant interaction. To examine whether we could enhance the capacity of $dTPI^{\Delta cat}$ to suppress $dTPI^{G74E}$, we designed a double-mutant aiming to revert heterodimer formation.

An allele was generated bearing both substitutions, $dTPI^{T73R,G74E}$, and animals homozygous for this allele displayed aggressive behavioral phenotypes and shortened lifespans (S4A, S4B and S4D Fig). Immunoprecipitation experiments found that the double substitution reduced $dTPI^{T73R,G74E}$ heterodimerization with $dTPI^{\Delta cat-CFP}$ relative to $dTPI^{G74E}$ (Fig 6). Finally, when paired with the $dTPI^{\Delta cat}$ allele, the addition of the T73R substitution to $dTPI^{G74E}$ enhanced the capacity for $dTPI^{\Delta cat}$ -behavioral complementation (S4A and S4B Fig). The mean time to



Fig 6. Heterodimerization of inactive TPI and dimer interface mutations. (A) $dTPI^{\Delta cat-CFP}$ interacts modestly with $dTPI^{M80T}$, $dTPI^{T73R}$, and $dTPI^{T73R,G74E}$, yet robustly with $dTPI^{G74E}$. Representative immunoprecipitation and input blots are shown with (B) IP:anti-GFP quantification n = 3. Quantification represents 25kD TPI IP signal, with negative control subtracted, normalized to the lysate β -tubulin loading control, and compared to WT. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, *ns* indicates no significance, ** p<0.01, and *** p<0.001.

doi:10.1371/journal.pgen.1005941.g006

recovery after mechanical stress was reduced from 204 sec. in $dTPI^{T73R,G74E}$ homozygotes to 50 sec. in $dTPI^{T73R,G74E}/dTPI^{\Delta cat}$ animals with approximately 60% of the animals no longer responding to the stressor (defined as a recovery time \leq 5 sec.) (S4A Fig); similar complementation was observed in the thermal stress assay (S4B Fig). Curiously, the longevity of the $dTPI^{T73R,G74E}/dTPI^{\Delta cat}$ animals was unchanged relative to $dTPI^{T73R,G74E}$ homozygotes (S4D Fig); this is the second time that TPI deficiency behavioral abnormalities and longevity have not paralleled each other [18], suggesting the possibility of independent pathogenic mechanisms (see Discussion).

The inverse correlation between $dTPI^{\Delta cat}$ heterodimerization and behavioral complementation suggested that $dTPI^{\Delta cat}$ did not complement TPI deficiency behavioral phenotypes via heterodimer formation. Further, disease severity did not correlate with isomerase activity (<u>S5B</u> Fig); complementation of the dimer-mutant alleles with $dTPI^{\Delta cat}$ failed to increase isomerase activity and in all but one case significantly decreased activity (<u>S4C</u> and <u>S5</u> Figs). These data led us to conclude that $dTPI^{\Delta cat}$ does not "suppress" TPI deficiency behavioral phenotypes through a general influence on TPI catalytic activity.

Severe TPI dimer impairment alters synaptic function

TPI deficiency complementation was not corroborated by an enhancement of TPI catalysis; however, the mean temperature-dependent time to paralysis of the $dTPI^{T73R}$ allele (27 sec) was a striking result (Fig 2B) and suggested a previously unknown role of TPI. Rapid (<60 sec.) temperature-dependent paralysis had only been identified in a handful of mutants in *Drosoph-ila* and typically results from neural conductance or synaptic vesicle recycling defects [45]. To determine whether TPI was influencing vesicle dynamics, we first examined vesicle endocytosis at the synapse using the lipophilic dye, FM1-43. FM1-43 is a water soluble membrane dye that increases its fluorescence when bound to cellular membranes. During endocytosis, the dye will bind to the outer leaflet of the plasma membrane and become internalized within the synapse providing an optical measurement of endocytosis. Measuring vesicle dynamics in this context allowed us to assess two possibilities; i) a primary recycling defect due to impaired endocytosis, or ii) a secondary recycling defect due to aberrant exocytosis.

We dissected larvae homozygous for *dTPI^{WT}*, *dTPI^{T73R}*, and *Shi^{ts1}* as previously detailed [46]. The NMJ preparations were heated to 38°C over 3 min. and a loading curve was generated from a series of three different high $[K^+]$ + FM1-43 loading times– 15 sec., 30 sec., and 60 sec. as previously detailed [47]. $dTPI^{WT}$ displays a progressive increase in dye loading from 15 sec. to 60 sec. (Fig 7A), while the temperature sensitive dynamin mutant control Shi^{ts1} showed no signs of vesicle recycling at any heated time points (Fig 7D, data not quantified). Conversely, although *dTPI*^{T73R} displayed similar loading to *dTPI*^{WT} at 15 and 30 sec., *dTPI*^{T73R} exhibited a striking 50% decrease in loading at 60 sec. (Fig 7A, 7B and 7D). This progressive decrease in endocytosis was stimulation and temperature dependent; loading experiments performed at room temperature did not exhibit an endocytic defect (Fig 7C). As previous experiments had highlighted the capacity of $dTPI^{Acat}$ to complement the adult behavioral defects of $dTPI^{T73R}$, we examined $dTPI^{T73R}/dTPI^{\Delta cat}$ larvae to assess the relationship between vesicle endocytosis and animal behavior. The $dTPI^{T73R}/dTPI^{\Delta cat}$ animals displayed a significant increase in vesicle endocytosis relative to $dTPI^{T73R}$ (Fig 7B and 7D). These results demonstrate that $dTPI^{\Delta cat}$ complements adult behavior and vesicle endocytosis defects. The utilization of chemical stimulation in these preparations demonstrates a synaptic defect arising from the severe dTPI^{T73R} dimer mutation as this methodology bypasses conductance requirements.

A reduction in vesicle dye uptake could be derived from defects in endocytosis or exocytosis, and indeed, these activities are intimately linked [48]. To examine temperature-dependent





Fig 7. *dTPI*^{T73R} **impairs NMJ synaptic vesicle dynamics.** (A) An FM1-43 timecourse at the NMJ with loading times of 15, 30, and 60 sec., (B) with quantification of 60 sec. at 38°C, and (C) 60 sec. at room temperature (RT). (D) Representative images of $dTPI^{WT}$, $dTPI^{T73R}$, $dTPI^{T73R/ACat}$ and Sh^{ts1} , n = 6. (E) FM1-43 unloading is unchanged between $dTPI^{WT}$ and $dTPI^{T73R}$ at 38°C with animal replicates indicated, and (F) representative images. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, ***p<0.001. Scale bars = 5µm.

changes in vesicle fusion, $dTPI^{WT}$ and $dTPI^{T73R}$ animals were i) loaded with dye at RT for 3 min., ii) washed with 0 mM Ca²⁺ HL-3, iii) imaged, iv) heated to 38°C, v) vesicle fusion initiated with 30 sec. of high [K⁺] HL-3, and vi) reimaged. Care was taken to ensure the same synapses were imaged at loading and unloading timepoints. Preliminary experiments demonstrated that 60 sec. of high [K⁺] stimulation completely unloaded the synapses in each genotype; therefore 30 sec. was analyzed to achieve a measurable dynamic range. Unloading experiments at elevated temperatures demonstrated no change in vesicle exocytosis between $dTPI^{WT}$ and $dTPI^{T73R}$ at 38°C (Fig 7E and 7F).

Finally, functional changes at the synapse can be the result of acute impairments in recycling machinery or more chronic developmental defects. Mutations in the E3- ubiquitin ligase *Highwire* or alterations in the *trans*-synaptic signaling proteins *wingless* and *Glass-bottom boat* have been shown to alter synaptic function through primary changes in development [49-51]. These changes in synaptic physiology are accompanied by dramatic alterations in synaptic morphology, a hallmark of neurodevelopmental defects. To examine whether aberrant neuro-development may contribute to this recycling deficit, we morphologically characterized the



Fig 8. TPI dimer interface substitutions do not alter NMJ development and morphology. (A) NMJ morphology of segment A2 muscle 6/7 was characterized for (B) bouton number and (C) branching. Boutons were defined as varicosities greater than 2 μ m in diameter. Neither parameter showed significant differences elicited by the mutations, relative to either $dTPI^{WT}$ or $dTPI^{WT-CFP}/TPI^{WT}$. CFP tags did not alter the behavioral deficits of the animals (S6 Fig). n = 10. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, *ns* indicated no significance. Scale bar = 10 μ m.

Drosophila NMJ from segment A2, muscle 6/7; this particular NMJ is highly elaborate and therefore sensitive to developmental perturbations. The $dTPI^{M80T}$, $dTPI^{T73R}$, and $dTPI^{G74E}$ alleles all exhibited early lethality if maintained at 25°C, therefore development was scored at RT. Third instar larva were dissected, and an assessment of bouton number and branches revealed no significant developmental differences in the thermal-stress sensitive mutants relative to $dTPI^{WT}$ (Fig 8). These results suggest that the synaptic defect is an acute disruption of function, and not likely a secondary defect caused by altered development. Collectively, these data demonstrate that TPI deficiency thermal-stress sensitivity is characterized by acute perturbation of synaptic vesicle dynamics.

Discussion

We conclude that the impairment of TPI dimerization is sufficient to elicit neurologic dysfunction. Dimerization of TPI is critical to its conformation, stability, and catalytic properties; and disruption of these molecular properties impedes vesicle dynamics at the synapse. Impaired synaptic function is thermal-stress dependent, and both vesicular and behavioral abnormalities can be genetically complemented through a catalytically inactive *TPI* allele.

Triosephosphate isomerase dimerization

The pathogenic hTPI^{M80T} substitution impairs TPI dimerization. These results were obtained from purified proteins and do not corroborate those from non-denaturing gel filtration experiments performed on animal lysates [26]. However, several *in vitro* studies have found that mutations that impair TPI dimerization severely destabilize the protein [28,30,44,52,53]. *In vivo*, unstable proteins are bound by chaperones and either refolded, targeted to the

proteasome, or aggregate [54]. The results presented here suggest that dTPI^{M80T} does not cause TPI to aggregate (Fig 3), while previous work extensively details the recruitment of Hsp70 and Hsp90 to dTPI^{M80T} and its degradation through the proteasome [34]. Therefore, we hypothesize that TPI monomer may not have been detected previously in animal lysates due to its rapid sequestration and degradation. We believe these data, along with the previous inability to identify monomer *in vivo*, collectively suggest that TPI does not stably exist *in vivo* as a soluble monomer.

We utilized our GE system to generate two additional *TPI* alleles with point mutations at the dimer interface that have previously been shown to impair homodimerization [28]. These substitutions were located at the tip of the 3rd loop of TPI that extends into its dimer partner and stabilizes/rigidifies a network of hydrophobic interactions and hydrogen bonds which form the dimer interface [28,29,55]. The substitution of these dimer interface residues resulted in severely pathogenic *TPI* alleles, eliciting greater behavioral dysfunction and shorter lifespans than $dTPI^{M80T}$ (Fig 2).

Molecular basis of TPI deficiency

A universal molecular mechanism of TPI deficiency pathogenesis is currently unclear. To date, two crystal structures of disease-associated TPI mutations have been reported [15,18]. These mutations were found in two distinct structural regions of the TPI homodimer. The first structure to be solved was from the most commonly diagnosed TPI deficiency substitution, hTPI^{E104D} [15]. The hTPI^{E104D} substitution is a conservative alteration of a charged residue at the dimer interface that results in reduced dimer stability, but unchanged catalytic activity [15,16]. The second disease-associated structure was an hTPI^{I170V} substitution, a conservative substitution found on the catalytic lid of the enzyme that enhances thermal stability and reduces catalytic activity [16,18]. The dimer substitutions used in this study share several molecular characteristics with the TPI^{E104D} human mutation, and given the conservation of the TPI enzyme, we propose that these patients likely share similar molecular and cellular dysfunction as identified in this study. Conversely, the dimer substitutions are not predicted to share many molecular similarities with the hTPI^{I170V} mutation indicating that dimerization defects are sufficient but not necessary to elicit TPI deficiency. It is interesting to note that although dTPI^{T73R} and hTPI^{1170V} both exhibit mechanical and thermal stress sensitivities in Drosophila, the behavioral dysfunction caused by the dimer substitutions is far more severe, and $hTPI^{II70V}$ does not influence animal longevity [18]. Further, the capacity to attenuate behavioral dysfunction but not longevity in $dTPI^{T73R,G74E}/dTPI^{\Delta cat}$ suggests an independent pathogenic mechanism (S4 Fig) that may not be determined by TPI dimerization.

TPI dimerization and protein stability *in vivo* will ultimately influence catalytic capacity, and many TPI activity measurements from animal models and patient tissue samples have identified a reduction in isomerase activity [7,14,17,18,25,56-58]. However, our measurements of isomerase activity from healthy and affected animal lysates argue that TPI dimer integrity is a stronger determinant of behavioral dysfunction. Still, this could indicate that reduced TPI activity is corollary to the disease or a contributing factor to an alternative pathogenic mechanism. Previous studies have demonstrated that the redox state is altered in TPI deficient cells and organisms [33,59,60]. The redox status in TPI deficiency is proposed to be altered by abnormal flux through the pentose-phosphate pathway as well as potential accumulation of advanced glycation end-products (AGEs) [16,20,33,59-61], and the accumulation of redox damage in the nervous system is strongly linked with several neurodegenerative diseases [62-64]. Interestingly, a study in yeast demonstrated differences in redox responses between the E104D and I170V mutations [16]; these results could imply different modes of pathogenesis that are dependent on the conformational and catalytic states of TPI.

hTPI^{∆cat} predicted heterodimers

One unresolved aspect of this study was the inability of $dTPI^{\Delta cat}$ to fully complement the behavioral defects of $dTPI^{G74E}$. Co-IP experiments suggested that complementation correlated with an inability of dimer-interface mutants to form heterodimers with $dTPI^{\Delta cat}$ (Fig 6). These data would imply that $dTPI^{G74E}$ may be exhibiting a dominant negative effect as a heterodimer, but this conclusion was inconsistent with our genetic analyses of $dTPI^{WT}/dTPI^{G74E}$ animals (Fig 4). To investigate whether TPI^{G74E} may be interacting differently with TPI^{WT} than with $TPI^{\Delta cat}$, we purified, crystallized, and determined the structure of $hTPI^{\Delta cat}$ at 1.7Å resolution.

While the hTPI^{K13M} substitution (hTPI^{Δ cat}) resulted in multiple rearrangements that mimic the open or non-catalytic TPI conformation, the dimeric interface remained essentially unchanged, including the peptide backbone of Loop3 and sidechain positions hT75, hG76, and hM82 (Fig 5). To address how hTPI^{T75R} and hTPI^{G76E} substitutions may influence the dimer interface, we generated models of heterodimers in which one subunit contained either hT75R, hG76E, or both hT75R and hG76E substitutions, while the other monomer remained unaltered. Models were made using either wild-type TPI hTPI^{WT} (PDB: 4POC) or the hTPI^{Δ cat} (PDB: 4ZVJ) as the structural template, and subjected to analysis by RosettaBackrub [65]. Briefly, Rosetta scores are predictions of the most energetically stable conformations with higher scores indicating less favorable positioning of the model. The algorithm was run 50 times for each mutation to be modeled. Of these 50 simulations, the lowest scores of hTPI^{Δ cat} :: hTPI^{T75R} and hTPI^{Δ cat}::hTPI^{G76E} were selected and shown (Fig 9A and 9B), while the modeled structures whose Rosetta scores fell within the best 10% of its respective ensemble were collected for analysis (Fig 9C).

Modeling the hTPI^{T75R} and hTPI^{G76E} substitutions as homodimers or heterodimers with the hTPI^{WT} structure produced high Rosetta scores, predicting poor energetic favorability (Fig 9C) in agreement with our gel filtration experiments (S2 Fig). To examine TPI^{Δ cat} heterodimers we used the new hTPI^{Δ cat} structure to model hTPI^{Δ cat}::hTPI^{T75R} and hTPI^{Δ cat}::hTPI^{G76E}. These experiments predicted a high Rosetta score for hTPI^{Δ cat}::hTPI^{T75R} and a very low one for hTPI^{Δ cat}::hTPI^{G76E}, corroborating the results of our animal lysate coIP experiments and suggesting the simulated heterodimers may accurately represent the conformations of these molecules.

The hTPI^{Δ cat}::hTPI^{T75R} heterodimer with the lowest Rosetta score suggests the hR75 residue may orient toward the catalytic pocket of hTPI^{Δ cat}, lining the floor of the substrate-binding pocket through the displacement of hE165 and hK13M (Fig 9A). This orientation of hR75 into the catalytic pocket is similar to that previously described by Wierenga and colleagues [28]. The hTPI^{Δ cat}::hTPI^{G76E} heterodimer with the lowest Rosetta score suggests that hE76 finds a stable position in the dimer interface through the displacement of hE104 and hR98, possibly via coordination of the terminal amide of hN65 (Fig 9B). Interestingly, perturbation of hE104 has been shown to significantly alter the TPI dimer interface and elicits TPI deficiency in humans through a conservative hTPI^{Δ cat}::hTPI^{G76E} dimer interface is drastically altered relative to hTPI^{Δ cat} homodimers.

Triosephosphate isomerase deficiency neuropathogenesis

Drosophila TPI deficiency neurologic dysfunction is characterized by impaired vesicle dynamics at the neuronal synapse, a defect we believe is likely conserved in human patients. The key to deciphering this pathogenic mechanism was the behavioral severity of the newly generated TPI dimer interface mutants. The *dTPI*^{T73R} allele exhibited temperature-dependent paralysis at a mean time of approximately 27 sec., an acute behavior that is rare and highly enriched for





Fig 9. hTPI^{Δcat} **models predict that hTPI**^{Δcat}**::hTPI**^{G76E} **heterodimerization alters the TPI dimer interface.** (A) The R75 side chain may insert itself into the active site in the context of a hTPI^{Δcat}::hTPI^{T75R} heterodimer. Shown is a superposition of hTPI^{WT} (grey), hTPI^{Δcat} (blue), and hTPI^{Δcat}::hTPI^{T75R} (green) obtained from Rosetta Backrub modeling. Loop3, containing T75R and G76E, is indicated in tan. For clarity, the mainchain of hTPI^{Δcat}::hTPI^{T75R} has been omitted. (B) Repositioning of E104 and R98 side chains assists the dimer interface to accommodate the G76E substitution in Rosetta modeling. Superposition of hTPI^{Δcat} structure (blue) with the hTPI^{Δcat}::hTPI^{G76E} model (green). In both structures, Loop 3 is shown in tan and selected active site pocket residues are indicated. (C) Modeling the effect of T75R and G76E substitutions in the context of hTPI^{WT} and hTPI^{Δcat} structures. Normalized scores for the best 10% of 50 simulations were averaged for each experimental structure with the indicated computational substitution. Higher scores indicate a resulting model that is less favorable using the Rosetta energy function.

synaptic or conductance defective mutants. Only a handful of *Drosophila* mutant alleles have been identified with rapid temperature-dependent paralysis, including those of voltage-gated Na⁺, K⁺, and Ca²⁺ channels (*para, sei, cac*) [66–68], the sodium-potassium exchanging ATPase (*ATPa*) [69], and components of vesicle fusion and recycling (N-ethylmaleimide sensitive factor-*dNSF1*, dynamin–*Shi*) [70,71]; and after noting these phenotypic similarities we broadly examined synaptic function. Stimulation was conducted using high [K⁺] bath applications, thereby bypassing the participation of Na⁺ and K⁺ channels. The *dTPI^{T73R}* mutants were characterized by normal endocytosis during acute stimulations (15 and 30 sec.) but exhibited a dramatic reduction after 60 sec. of stimulation with the *dTPI^{Acat}* allele significantly increased the temperature-dependent FM1-43 loading at these terminals (Fig 7B and 7D), similar to its complementation of adult thermal stress-induced paralysis. Finally, measurements of vesicle exocytosis at these elevated temperatures did not indicate an exocytic defect (Fig 7E and 7F). We were unable to detect terminal signal above background after 60 sec. of unloading, so we cannot unequivocally define the nature of the vesicular dysfunction, but the data suggest it is likely due to impaired endocytosis.

The observed synaptic defect provides insight into why the majority of human patients present with TPI mutations affecting the dimer interface. First, all substitutions that disrupt the dimer interface have been shown to destabilize the enzyme in vitro [28,29,55]. This destabilization is likely responsible for the reduced cellular TPI found in patient samples, and our work with the *dTPI^{M80T}*, *dTPI^{T73R}*, and *dTPI^{G74E}* mutants provide additional evidence that dimer interface substitutions reduce TPI levels in vivo (Fig 3A and 3B). Secondly, the cellular depletion of dTPI^{M80T} has been shown to be mediated by heat shock protein sequestration and proteasomal degradation [34]. If chaperones sequestered and degraded these misfolded or unstable proteins, this would likely prevent the distribution and maintenance of TPI at specific subcellular locales. Recent work has shown that the anterograde transport of globular/soluble proteins to the terminals is a slow process, moving at a rate of approximately 0.008–0.01µm/ sec [72]. To put this in the context of the *Drosophila* nervous system, the length of the relatively short larval motor axon innervating muscle 4 of segment A3 has been measured to be ~220µm [73]. Based on these approximations, one could estimate that it would take ~6 hrs for TPI translated in the soma to be transported to the axonal terminal. In this way, substitutions that affect protein stability would likely result in improper localization or sequestration of TPI during distal transport, ultimately depleting TPI at the synapse. This proposal is also consistent with the inability of RNAi knockdown to recapitulate Drosophila TPI deficiency behavioral phenotypes, as RNAi alters mRNA transcript levels rather than enzyme conformation or stability. RNAi knockdown of TPI^{WT} would reduce, but still allow the transport and stable accumulation of TPI^{WT} at the synaptic terminal.

Two other *Drosophila* glycolytic mutants, aldolase and phosphoglycerate kinase, have been shown to exhibit temperature-dependent paralysis though with longer onsets [74,75]. The phosphoglycerate kinase mutant was shown to exhibit synaptic dysfunction, and the authors asserted that an inhibition of vesicle recycling was likely the cause of the functional defect [75]. In both cases, the animals were found to have depletions in lysate ATP [74,75]. It is attractive to speculate that all three of these glycolytic mutants may suggest a pivotal role for glycolysis within synaptic vesicular dynamics, and indeed, recent measurements of ATP consumption in the synapse suggest that glycolytic ATP is the primary substrate used to support synaptic function [76,77]. However, the role of glycolytic proteins and their putative energetic importance at the synapse is controversial. Many research groups assert the preeminent utilization of mitochondrial ATP at these sites [78–81], while the lactate shuttle hypothesis largely circumvents a role for neuronal glycolysis [82,83]. Further, the absence of a correlation between TPI catalytic activity and behavioral phenotypes suggests that the enzyme may be complexing with another molecule to facilitate synaptic vesicle cycling.

How the dimerization or integrity of the TPI dimer interface impacts synaptic vesicle dynamics remains a mystery, though one candidate for a molecular complex is the actin-regulatory protein cofilin. In *Drosophila*, cofilin (*twinstar*) and *twinfilin* mutants have been demonstrated to elicit functional and developmental neurologic defects [84–86], and actin-regulatory proteins such as cofilin, actin-depolymerizing protein (ADP), and twinfilin are known to influence synaptic vesicle dynamics [85,87,88]. Recently, cofilin was found to bind to TPI in both its inactive and active forms [89]. The precise binding site between cofilin and TPI is unknown, though with its mixture of charged and hydrophobic pockets, the TPI dimer interface may provide a suitable site for this interaction. Additional studies will be needed to specifically delineate the role of TPI in the synapse.

In conclusion, to our knowledge this work is the first to highlight a critical role for TPI in the cycling of vesicles at the synapse, with behavioral correlates similar to the inactivation of vesicle fusion/recycling proteins. These observations help clarify the neurologic symptoms seen in patients and will direct future therapeutic strategies. The findings of this study will guide future investigations regarding the contribution of TPI localization and function to synaptic vesicle dynamics, and ultimately how these properties are perturbed in TPI deficiency.

Materials and Methods

Animal strains

The Vienna *Drosophila* RNAi Center (VDRC) line used for knockdown experiments was stock #25644 [37]; experiments were also conducted with #25643 with similar results. The *w;actin-GAL4*, *UAS-GAL4*; animals were generated by recombining the second chromosomes of the *Drosophila* Genetic Resource Center (DGRC) stock $y^{I} w^{I118}$; $P(w^{+mC} = UAS-Gal4.H)12B$ and Bloomington Stock Center stock $y^{I}w^{*}$; $P(Act5C-GAL4)25FO1/CyO,y^{+}$; recombinants were screened molecularly and balanced in a w^{I118} background. The following TPI alleles in this study were generated using the GE system: $dTPI^{WT}$, $hTPI^{WT}$, $dTPI^{WT-CFP}$, $dTPI^{M80T}$, $hTPI^{M82T}$, $dTPI^{T73R}$, $dTPI^{T73R-CFP}$, $dTPI^{G74E}$, $dTPI^{G74E-CFP}$, $dTPI^{T73R,G74E}$, $dTPI^{Acat}$, and $dTPI^{Acat-CFP}$. The development of the GE system and the production of the $dTPI^{WT}$, $dTPI^{M80T}$, $dTPI^{Acat}$, and $dTPI^{Acat-CFP}$ alleles were initially described elsewhere [25]. Briefly, GE involves the replacement of the TPI gene locus with a phiC31 integration site through homologous recombination. The phiC31 integration system allows nearly seamless integration of complementary vector constructs directly into the TPI gene locus to maintain endogenous spatial and temporal regulation. The TPI^{mul} allele used in <u>S1 Fig</u> was generated previously, formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19]. RNAi experiments used the TPI^{M80T} allele formerly known as TPI^{S10} [19].

This study uses the established nomenclature for TPI, assuming the start methionine is removed following translation [13]; all residue numbering in this study uses the same convention. An alignment is included for clarification (Fig 1A). All animal populations assessed were approximately equivalent mixtures of males and females.

Mutagenesis and genomic engineering

Site directed mutagenesis was performed using the QuikChange Lightening Site-Directed Mutagenesis Kit (Agilent Technologies). Mutagenesis primers were generated (Integrated DNA Technologies) to introduce a Thr-to-Arg codon change at position 73, and a Gly-to-Glu change at position 74 –both separately and together for the purpose of creating the double-mutant. Mutagenesis was performed on the previously published *pGE-attBTPI*⁺ plasmid [25] and confirmed by sequencing. Once the constructs were generated, *TPI* GE was performed using previously published methods [25,90,91]. Briefly, the *PGX-TPI* founder animals were mated to *vasa-phiC31*^{ZH-2A} animals expressing the integrase on the X chromosome and their progeny injected with *pGE-attBTPI* constructs. Integration events were identified via the w^+ phenotype and verified molecularly.

Human TPI enzyme purification

Human TPI enzyme was purified as outlined previously [18].

Dynamic Light Scattering

DLS measurements were taken using a DynaPro Plate reader (Wyatt Technology) equipped with a temperature control unit. Purified BSA (Sigma Aldrich), hTPI⁺, and hTPI^{M80T} were

diluted to concentrations of 3.75μ M, 15μ M and 30μ M in 100 mM triethanolamine (TEA); pH 7.6. Three 75 μ l aliquots were loaded onto a 384-well microplate and read at 37°C. Ten measurements were taken per sample and Dynamics V6 software (Wyatt Technology) was used to process the scattering data, generating autocorrelation functions. Autocorrelation functions were then analyzed to obtain the hydrodynamic radii. Student's T test was used to compare samples. DLS experiments were performed three times.

Gel filtration chromatography

Gel filtration was performed as outlined previously [26]. Purified TPI samples were diluted to 15 μ M in mobile phase, 100 μ l were injected in triplicate, and their elution monitored at 280 nm. Experiments were performed three times. Chromatography traces were collected and analyzed using EZStart 7.3 (Shimadzu) to quantify the relative monomer and dimer populations. Curve integration data were compared using Student's T test.

TPI enzyme assays

Isomerase activity was determined using an NADH-linked assay as previously detailed [25,92]. Lysates were diluted to 0.1 μ g/ μ l in 100 mM TEA pH 7.6 + inhibitors and enzyme activity was assessed. Reaction assays were performed in triplicate using 80 μ l mixtures composed of 0.5 mM NADH, 0.752 mM GAP, 1 unit glycerol-3-phosphate dehydrogenase and 1 μ g of lysate protein in 100 mM TEA; pH 7.6. Consumption of NADH was monitored at 340 nm and 25°C using a SpectraMax Plus 384 microplate reader (Molecular Devices). All reactions were performed at least three times. Reaction components were purchased from Sigma-Aldrich. Enzyme activity curves were normalized to reactions performed without GAP. A one-way ANOVA was performed to assess variance, and data sets were compared using Tukey's posthoc analysis.

Behavioral testing and lifespan analysis

Mechanical stress sensitivity was examined on Day 1 by vortexing the animals in a standard media vial for 20 seconds and measuring time to recovery, similar to [93]. Briefly, recovery is defined as two purposeful WT movements including righting, grooming, climbing, or walking. Thermal stress sensitivity was assessed on Day 2 by acutely shifting animals to 38°C and measuring time to paralysis, as previously described [24,69]. In these assays, the animals typically seized and either flipped over onto their backs or fell sideways with no successive coordinated movements, i.e. righting, climbing, walking, grooming. Behavioral responses were capped at 360 and 600 seconds where indicated and reported as 360 and 600 sec. Animal lifespan determinations were performed at 25°C as previously described [69]. All assays used approximately equivalent numbers of males and females. One-way ANOVAs were performed with Tukey's post-hoc analysis to compare behavior, and lifespans were assessed with Log-rank (Mantel-Cox) survival tests.

Immunoblots

Animals were collected and aged 1–2 days at room temperature. Ten fly heads were obtained in triplicate from each genotype and processed as described previously [34]. Blots were incubated with anti-TPI (1:5000; rabbit polyclonal FL-249; Santa Cruz Biotechnology), anti-ATPalpha (1:10,000; mouse monoclonal alpha5; Developmental Studies Hybridoma Bank), or anti- β tubulin (1: 6,000; rabbit polyclonal H-235; Santa Cruz Biotechnology). Densitometric analyses of the scanned films were performed on unsaturated exposures using ImageJ software available from the National Institutes of Health. A one-way ANOVA was performed to assess variance of TPI levels and data sets were compared using Tukey's post-hoc analysis.

Filter-trap dot blot

The filter-trap dot blot was modified from methods published previously [36]. Animals were aged 1–2 days, collected and homogenized in 1X PBS (2.7 mM KCl, 137 mM NaCl, 2 mM NaH₂PO₄, 10 mM Na₂HPO₄; pH 7.4) supplemented with cOmplete mini Protease Inhibitors (Roche Diagnostics), and diluted to 1 μ g/ μ l; wells were loaded as indicated. Samples were diluted 1:2 in 1% SDS, 1X PBS, boiled for 5 min., and filtered through a cellulose acetate membrane (Whatman, 0.2 μ m pore) using a 96-well vacuum dot blot apparatus. Positive controls were collected from PC12 cells transfected with a construct expressing huntingtin "exon1" bearing a stretch of 97 glutamines and C-terminally tagged with EGFP. The membrane was washed four times with 1% SDS-PBS, blocked with Odyssey Blocking Buffer (LiCor), and primary antibodies applied in Odyssey Blocking Buffer. Blots were incubated with anti-TPI (1: 5000) and anti-GFP (1: 5000; rabbit polyclonal; FL; Santa Cruz Biotechnology). The membranes were then washed and incubated with the secondary antibody IRDye 800-conjugated goat anti-rabbit (LiCor) at 1: 20,000 in Odyssey Blocking Buffer. Direct-to-scanner detection and blot visualization were performed using a LiCor Odyssey scanner. Filter-trap experiments were performed twice.

Coimmunoprecipitation experiments

Coimmunoprecipitation was performed using the Pierce Co-Immunoprecipitation Kit (Thermo Scientific) as per manufacturer's instructions. Lysates were generated by mechanically homogenizing 50 animals in 0.5 ml of IP Lysis buffer (25 mM Tris, 150 mM NaCl, 1 mM EDTA, 1% NP-40, 5% glycerol; pH 7.4) supplemented with cOmplete mini Protease Inhibitors. After homogenization, lysates were frozen in liquid nitrogen, thawed, then centrifuged twice at 5,000 *g* to pellet exoskeletal debris. Supernatants were collected and diluted to 1 μ g/ μ l, and 400 μ g were loaded onto 25 μ l of gel pre-coupled with 10 μ g of anti-GFP. A negative control was performed using uncoupled gel and $dTPI^+/dTPI^{Acat}$ lysate. Samples were incubated overnight at 4°C and washed ten times with IP Lysis buffer at 4°C. Beads were eluted by boiling with 70 μ l of 2X SDS–PAGE sample buffer, separated via SDS-PAGE, immunoblotted, and analyzed as outlined above. Coimmunoprecipitations were performed three times.

FM1-43 imaging experiments

Images were taken with an Olympus BX51WI fluorescence microscope with Till Photonics Polychrome V monochromator excitation, and Hamamatsu C4742-95 digital camera. Heterozygous TPI^{T73R} larvae were maintained over TM6B, and Tb^+ 3rd-instar larvae selected for analysis. Dissection and preparation of larval NMJs were performed as described previously [46]. FM1-43FX dye [Molecular Probes, Invitrogen] loading was performed as previously detailed [47]. Briefly, animals were dissected in ice cold 0 mM Ca²⁺ HL-3 with 0.5 mM EGTA, motor neurons severed, and the preps heated to room temperature or 38°C over the course of 3 min. Bath temperature was monitored throughout the experiments with a microthermal probe to ensure consistency [Fisher Scientific]. Loading experiments were performed with room temperature or 38°C preheated 90 mM KCl, 1.5 mM CaCl₂ HL-3 supplemented with 4 μ M FM1-43FX, and preparations were washed quickly and thoroughly during the experiments to avoid Ca²⁺ chelation. After loading, preparations were washed with 15 ml of 0mM Ca²⁺ HL-3 with 0.5 mM EGTA at room temperature for 10 min. Unloading experiments were performed as follows: preparations were loaded for 3 min at room temperature; washed with 15 ml of 0mM Ca² ⁺ HL-3 with 0.5 mM EGTA at room temperature; imaged; washed with 38°C 0 mM Ca²⁺ HL-3; heated to 38°C over 3 min.; unloaded using 38°C 90 mM KCl, 1.5 mM CaCl₂ HL-3 for 30 sec.; and the same synapse was imaged again.

Preparations were imaged with a water immersion 60X objective, using 450 nm excitation and a 500 nm longpass filter [Chroma Technology]. Simple PCI imaging software was used for acquisition and ImageJ for analysis. Two NMJs from muscles 6/7 were assessed per animal, one from segment A2 and A3. Six animals were assessed per genotype per time point for a total of 12 NMJs per experimental condition. After acquisition, images were relabeled by an independent researcher and blinded analysis was performed on the raw images. Boutons were outlined and intensity measured, with background subtracted from adjacent tissue. Pair-wise analyses were performed using a two-tailed Student's t test, while comparisons among multiple experimental conditions were performed using a one-way Analysis of Variance (ANOVA) with Tukey's post-hoc analysis.

NMJ morphological analyses

For NMJ morphological analyses, 3rd-instar larvae were collected and dissected as detailed above without transection of the descending motor neurons. Preparations were fixed in 3.5% paraformaldehyde HL-3, permeabilized with 0.1% Triton X-100 in 1X PBS (PBST), and blocked with 0.2% BSA in PBST (PBSTB) for 2 hrs at room temperature. Preps were washed and incubated with goat anti-HRP [Jackson Laboratories] at 1: 200 in PBSTB for 2 hrs at room temperature. Primary antibodies were removed, washed in PBSTB, and incubated with Cy3-labeled donkey anti-goat in PBSTB at 1: 400 for 1.5 hrs at room temperature. Preps were washed, mounted in VectaShield [Vector Laboratories], and imaged within three days. Images were acquired with an Olympus confocal FV1000 microscope, using a 559 nm excitation laser. Z stacks of segment A2 of muscle 6/7 were taken using 1 µm steps. The Z stacks were merged using Olympus FV1000 Fluoview Viewer, and morphology determined. Ten animals were assessed per genotype, one NMJ per animal, for a total of ten NMJs per experimental condition. Boutons were defined as varicosities at least 2 µm in diameter, and branches defined as extensions containing at least 2 boutons. Images were relabeled by an independent researcher for blinded analysis. Variance within the data set was examined using a one-way ANOVA, with comparisons made using Tukey's post hoc test.

Image analysis and presentation

All image quantification was performed on raw image files acquired below saturation. Representative images were selected on the basis of raw image measurements, and post-acquisition processing was performed uniformly with grouped images in parallel using ImageJ; in agreement with published guidelines [94].

Crystallization and structure determination of $hTPI^{\Delta cat}$

Recombinant hTPI^{Δ cat} containing the K13M substitution was expressed and purified as previously described [18] using affinity, anion exchange, and size exclusion chromatography. Purified protein was dialyzed into a buffer containing [20 mM Tris pH 8.8, 25 mM NaCl, 2.0% glycerol and 1 mM β -mercaptoethanol], and concentrated to 6 mg/ml prior to crystallization. Crystals of TPI^{Δ cat} were obtained using the vapor diffusion method with sitting drops containing 1 µl of protein and 2 µl of well solution [28–34% PEG2000 MME, 50 mM KBr]. Initial crystals grew within 3 days and were improved by successive rounds of microseeding. Crystals were cryoprotected in 40% PEG 2000MME, 20% glycerol, 50 mM KBr, prior to flash freezing in liquid nitrogen.

Data collection was performed at the National Synchrotron Light Source at beamline X25 and using a Pilutas 6M detector. Diffraction data was integrated, scaled, and merged using HKL2000 [95]. hTPI^{Δ cat} crystals belong to space group P2₁2₁2₁ and contain a dimer in the asymmetric unit. Initial phases were estimated for hTPI^{Δ cat} via molecular replacement using a previously determined structure of wild-type as our search model [18]. Model bias was reduced through simulated annealing and the model was further improved by manual model building combined with positional and anisotropic B-factor refinement within Phenix [96]. Model quality was validated using MolProbity [97]. Figs were generated using PyMOL (PyMOL Molecular Graphics System, Schrödinger, LLC). Coordinates and structure factors for hTPI^{Δ cat} have been deposited within the Protein Databank under accession code 4ZVJ.

Molecular modeling of dimer interface substitutions

The effect of hT75R and hG76E substitutions were modeled onto hTPI^{Δ cat} or hTPI^{WT} structural templates using the RosettaBackrub analysis as implemented within the RosettaBackrub server [98]. For all predictions, Rosetta version 3.1 was used as the algorithm with a backrub radius of 15 Å to ensure that perturbations extending away from the site of the substitution could be sampled. An ensemble of 50 structures was predicted for each TPI model, and their reported Rosetta scores normalized to the starting template. Structures whose Rosetta scores were in the most favorable 10% of the ensemble were used in Fig 9D.

TPI nomenclature

Triosephosphate isomerase research has been split between those studying the enzymatic and structural properties of the protein, and those studying its role in disease. Researchers focusing on the pathology of TPI deficiency typically use the abbreviation "TPI", whereas enzymologists and structural biologists used the abbreviation "TIM". The aim of this study was to determine the molecular mechanism of a disease mutation, and as such we have used the abbreviation "TPI".

Supporting Information

S1 Fig. *hTPI*^{M82T/null} **phenocopies the behavioral and longevity effects of** *dTPI*^{M80T/null}. (A) $dTPI^{M80T/null}$ and $hTPI^{M82T/null}$ elicit significant delays in time to recovery from mechanical stress at Day 3, (B) time to paralysis after thermal stress at Day 4, and (C) longevity compared to $dTPI^{WT/null}$ and $hTPI^{WT/null}$, respectively. All animals were reared at 25°C, and thermal stress paralysis times at 600 sec. represent wild type behavior; the assay was stopped at 10 min. as previously outlined for hTPI alleles [18]. n≥20 for behavior and n≥70 for lifespans. The TPI^{null} and $hTPI^{WT}$ alleles were generated previously [18,19], with TPI^{null} representing a deletion of two of three constitutive TPI exons. Comparisons were made with Student's T test, and lifespans by a Log-rank (Mantel-Cox) survival test, * indicates p<0.05 and *** p<0.001. (EPS)

S2 Fig. TPI dimer interface substitutions impair enzyme dimerization. Purified hTPI^{Δ cat}, hTPI^{T75R}, hTPI^{G76E}, and hTPI^{T75R,G76E} were examined via size-exclusion chromatography for monomer and dimer content. Each dimer interface substitution displays a significant increase in monomeric TPI relative to the enzyme dimer (n \geq 3). hTPI^{WT} and hTPI^{Δ cat} intermittently displayed some monomer species, and were prepared and analyzed in parallel with the dimer mutations. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, * indicates p<0.001.

(EPS)

S3 Fig. RNAi knockdown of TPI^{WT} fails to recapitulate $dTPI^{M80T}$ phenotypes. (A) TPI protein levels were knocked down ubiquitously and confirmed in thorax and (B) head tissues ($n \ge 3$) with quantification and representative images. (C) Knockdown animals failed to display typical mechanical-stress dependent paralysis at Day 5 aged at 25°C ($n \ge 15$). TPI^{WT} animals are w^{1118} ;; controls while the TPI^{M80T} allele used was formerly known as TPI^{sgk} [19]. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, *ns* indicates no significance, ** p<0.01, and *** p<0.001 relative to WT. Experiments shown are derived from knockdown using VDRC line #25644; VDRC #25643 was also examined, with similar results. (EPS)

S4 Fig. *TPI*^{*T73R,G74E*} **behavioral dysfunction and longevity.** (A) $dTPI^{T73R,G74E}$ is characterized by severe mechanical and (B) thermal stress sensitivity, which respond positively to complementation by $dTPI^{Acat}$. Thermal stress paralysis times at 360 sec. represent wild type behavior; the assay was stopped at 6 min. (C) $TPI^{T73R,G74E}$ exhibits reduced lysate catalytic activity compared to WT. (D) TPI^{Acat} fails to complement $TPI^{T73R,G74E}$ longevity. n \geq 30 for all behavior, n \geq 80 for all lifespans. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, and lifespans by a Log-rank (Mantel-Cox) survival test, *ns* indicates no significance, ** p<0.01, *** p<0.001. (EPS)

S5 Fig. Lysate isomerase activity does not predict disease presence or severity. (A) Lysate activity is indicated and (B) expanded as per the dashed box. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, *ns* indicates no significant differences and *** p<0.001. Biological replicates are indicated. (EPS)

S6 Fig. C-terminal CFP tags do not alter TPI dimer mutant neuropathology. C-terminal CFP tags were added to $dTPI^{T73R}$, $dTPI^{G74E}$, and $dTPI^{T73R,G74E}$. (A) Animals were crossed with similar alleles lacking the CFP tag, and the resulting progeny retain the mechanical- and (B) thermal-stress sensitivity. n>10. Thermal stress paralysis times at 360 sec. represent wild type behavior; the assay was stopped at 6 min. Comparisons were made with a One-way ANOVA using Tukey's post hoc test, *** indicates p<0.001. # indicates the animals did not paralyze within 360 sec. (EPS)

EPS)

Acknowledgments

The authors would like to thank Atif Towheed, Dinara Bulgari, Alicia Celotto, Kenneth Drombosky, Irene Arduini, Ronald Wetzel, and Karunakar Kar for their helpful discussions and technical assistance.

Author Contributions

Conceived and designed the experiments: BPR CGA AH ESL APV MJP. Performed the experiments: BPR AMZ CGA SBL ADT KAS AH. Analyzed the data: BPR CGA AMZ SBL ADT ESL APV KAS MJP. Wrote the paper: BPR APV MJP.

References

 Schneider AS. Triosephosphate isomerase deficiency: historical perspectives and molecular aspects. Baillieres Best Pract Res Clin Haematol. 2000; 13(1):119–40. Epub 2000/08/05. doi: S1521692600900616 [pii]. PMID: 10916682.

- Miwa S, Fujii H. Molecular basis of erythroenzymopathies associated with hereditary hemolytic anemia: tabulation of mutant enzymes. Am J Hematol. 1996; 51(2):122–32. Epub 1996/02/01. doi: <u>10.1002/</u> (SICI)1096-8652(199602)51:2<122::AID-AJH5>3.0.CO;2-# [pii] 10.1002/(SICI)1096-8652(199602) 51:2<122::AID-AJH5>3.0.CO;2-#. PMID: <u>8579052</u>.
- Climent F, Roset F, Repiso A, Perez de la Ossa P. Red cell glycolytic enzyme disorders caused by mutations: an update. Cardiovasc Hematol Disord Drug Targets. 2009; 9(2):95–106. Epub 2009/06/13. PMID: <u>19519368</u>.
- Orosz F, Olah J, Ovadi J. Triosephosphate isomerase deficiency: new insights into an enigmatic disease. Biochim Biophys Acta. 2009; 1792(12):1168–74. Epub 2009/09/30. doi: S0925-4439(09)00223-3 [pii] doi: 10.1016/j.bbadis.2009.09.012 PMID: 19786097.
- Sarper N, Zengin E, Jakobs C, Salomons GS, Mc Wamelink M, Ralser M, et al. Mild hemolytic anemia, progressive neuromotor retardation and fatal outcome: a disorder of glycolysis, triose- phosphate isomerase deficiency. Turk J Pediatr. 2013; 55(2):198–202. Epub 2013/11/07. PMID: 24192681.
- Wilmshurst JM, Wise GA, Pollard JD, Ouvrier RA. Chronic axonal neuropathy with triosephosphate isomerase deficiency. Pediatr Neurol. 2004; 30(2):146–8. Epub 2004/02/27. doi: <u>10.1016/S0887-8994</u> (03)00423-5 S0887899403004235 [pii]. PMID: <u>14984912</u>.
- Schneider AS, Valentine WN, Hattori M, Heins HL Jr. Hereditary Hemolytic Anemia with Triosephosphate Isomerase Deficiency. N Engl J Med. 1965; 272:229–35. Epub 1965/02/04. doi: <u>10.1056/</u> <u>NEJM196502042720503</u> PMID: <u>14242501</u>.
- Hollan S, Fujii H, Hirono A, Hirono K, Karro H, Miwa S, et al. Hereditary triosephosphate isomerase (TPI) deficiency: two severely affected brothers one with and one without neurological symptoms. Hum Genet. 1993; 92(5):486–90. Epub 1993/11/01. PMID: <u>8244340</u>.
- Aissa K, Kamoun F, Sfaihi L, Ghedira ES, Aloulou H, Kamoun T, et al. Hemolytic anemia and progressive neurologic impairment: think about triosephosphate isomerase deficiency. Fetal Pediatr Pathol. 2014; 33(4):234–8. Epub 2014/05/21. doi: 10.3109/15513815.2014.915365 PMID: 24840153.
- Serdaroglu G, Aydinok Y, Yilmaz S, Manco L, Ozer E. Triosephosphate isomerase deficiency: a patient with Val231Met mutation. Pediatr Neurol. 2011; 44(2):139–42. Epub 2011/01/11. doi: S0887-8994(10) 00408-X [pii] doi: 10.1016/j.pediatrneurol.2010.08.016 PMID: 21215915.
- Poll-The BT, Aicardi J, Girot R, Rosa R. Neurological findings in triosephosphate isomerase deficiency. Ann Neurol. 1985; 17(5):439–43. Epub 1985/05/01. doi: <u>10.1002/ana.410170504</u> PMID: <u>4004168</u>.
- Ovadi J, Orosz F, Hollan S. Functional aspects of cellular microcompartmentation in the development of neurodegeneration: mutation induced aberrant protein-protein associations. Mol Cell Biochem. 2004; 256–257(1–2):83–93. Epub 2004/02/24. PMID: 14977172.
- Orosz F, Olah J, Ovadi J. Triosephosphate isomerase deficiency: facts and doubts. IUBMB Life. 2006; 58(12):703–15. Epub 2007/04/12. doi: 10.1080/15216540601115960 PMID: 17424909.
- Orosz F, Olah J, Alvarez M, Keseru GM, Szabo B, Wagner G, et al. Distinct behavior of mutant triosephosphate isomerase in hemolysate and in isolated form: molecular basis of enzyme deficiency. Blood. 2001; 98(10):3106–12. Epub 2001/11/08. PMID: <u>11698297</u>.
- Rodriguez-Almazan C, Arreola R, Rodriguez-Larrea D, Aguirre-Lopez B, de Gomez-Puyou MT, Perez-Montfort R, et al. Structural basis of human triosephosphate isomerase deficiency: mutation E104D is related to alterations of a conserved water network at the dimer interface. J Biol Chem. 2008; 283 (34):23254–63. Epub 2008/06/20. doi: M802145200 [pii] doi: <u>10.1074/jbc.M802145200</u> PMID: <u>18562316</u>.
- Ralser M, Heeren G, Breitenbach M, Lehrach H, Krobitsch S. Triose phosphate isomerase deficiency is caused by altered dimerization—not catalytic inactivity—of the mutant enzymes. PLoS One. 2006; 1: e30. Epub 2006/12/22. doi: <u>10.1371/journal.pone.0000030</u> PMID: <u>17183658</u>; PubMed Central PMCID: PMC1762313.
- Hollan S, Magocsi M, Fodor E, Horanyi M, Harsanyi V, Farkas T. Search for the pathogenesis of the differing phenotype in two compound heterozygote Hungarian brothers with the same genotypic triose-phosphate isomerase deficiency. Proc Natl Acad Sci U S A. 1997; 94(19):10362–6. Epub 1997/09/18. PMID: <u>9294216</u>; PubMed Central PMCID: PMC23368.
- Roland BP, Amrich CG, Kammerer CJ, Stuchul KA, Larsen SB, Rode S, et al. Triosephosphate isomerase I170V alters catalytic site, enhances stability and induces pathology in a Drosophila model of TPI deficiency. Biochim Biophys Acta. 2015; 1852(1):61–9. Epub 2014/12/03. doi: S0925-4439(14)00316-0 [pii] doi: 10.1016/j.bbadis.2014.10.010 PMID: 25463631; PubMed Central PMCID: PMC4268122.
- Celotto AM, Frank AC, Seigle JL, Palladino MJ. Drosophila model of human inherited triosephosphate isomerase deficiency glycolytic enzymopathy. Genetics. 2006; 174(3):1237–46. Epub 2006/09/19. doi: genetics.106.063206 [pii] doi: <u>10.1534/genetics.106.063206</u> PMID: <u>16980388</u>; PubMed Central PMCID: PMC1667072.

- Gnerer JP, Kreber RA, Ganetzky B. wasted away, a Drosophila mutation in triosephosphate isomerase, causes paralysis, neurodegeneration, and early death. Proc Natl Acad Sci U S A. 2006; 103 (41):14987–93. Epub 2006/09/30. doi: 0606887103 [pii] doi: 10.1073/pnas.0606887103 PMID: 17008404; PubMed Central PMCID: PMC1581428.
- Ganetzky B, Loughney K, Wu CF. Analysis of mutations affecting sodium channels in Drosophila. Ann N Y Acad Sci. 1986; 479:325–37. Epub 1986/01/01. PMID: 2433999.
- Wu CF, Ganetzky B. Neurogenetic studies of ion channels in Drosophila. Ion Channels. 1992; 3:261– 314. Epub 1992/01/01. PMID: <u>1330057</u>.
- Wu CF, Ganetzky B, Jan LY, Jan YN, Benzer S. A Drosophila mutant with a temperature-sensitive block in nerve conduction. Proc Natl Acad Sci U S A. 1978; 75(8):4047–51. Epub 1978/08/01. PMID: 211514; PubMed Central PMCID: PMC392928.
- Palladino MJ, Hadley TJ, Ganetzky B. Temperature-sensitive paralytic mutants are enriched for those causing neurodegeneration in Drosophila. Genetics. 2002; 161(3):1197–208. Epub 2002/07/24. PMID: 12136022; PubMed Central PMCID: PMC1462168.
- Roland BP, Stuchul KA, Larsen SB, Amrich CG, Vandemark AP, Celotto AM, et al. Evidence of a triosephosphate isomerase non-catalytic function crucial to behavior and longevity. J Cell Sci. 2013; 126(Pt 14):3151–8. Epub 2013/05/04. doi: jcs.124586 [pii] doi: <u>10.1242/jcs.124586</u> PMID: <u>23641070</u>; PubMed Central PMCID: PMC3711204.
- Seigle JL, Celotto AM, Palladino MJ. Degradation of functional triose phosphate isomerase protein underlies sugarkill pathology. Genetics. 2008; 179(2):855–62. Epub 2008/05/07. doi: genetics.108.087551 [pii] doi: <u>10.1534/genetics.108.087551</u> PMID: <u>18458110</u>; PubMed Central PMCID: PMC2429879.
- Wierenga RK, Kapetaniou EG, Venkatesan R. Triosephosphate isomerase: a highly evolved biocatalyst. Cell Mol Life Sci. 2010; 67(23):3961–82. Epub 2010/08/10. doi: <u>10.1007/s00018-010-0473-9</u> PMID: 20694739.
- Schliebs W, Thanki N, Jaenicke R, Wierenga RK. A double mutation at the tip of the dimer interface loop of triosephosphate isomerase generates active monomers with reduced stability. Biochemistry. 1997; 36(32):9655–62. Epub 1997/08/12. doi: <u>10.1021/bi963086a</u> bi963086a [pii]. PMID: <u>9245397</u>.
- Schliebs W, Thanki N, Eritja R, Wierenga R. Active site properties of monomeric triosephosphate isomerase (monoTIM) as deduced from mutational and structural studies. Protein Sci. 1996; 5(2):229–39. Epub 1996/02/01. doi: <u>10.1002/pro.5560050206</u> PMID: <u>8745400</u>; PubMed Central PMCID: PMC2143345.
- Borchert TV, Kishan KV, Zeelen JP, Schliebs W, Thanki N, Abagyan R, et al. Three new crystal structures of point mutation variants of monoTIM: conformational flexibility of loop-1, loop-4 and loop-8. Structure. 1995; 3(7):669–79. Epub 1995/07/15. PMID: 8591044.
- Guix FX, III-Raga G, Bravo R, Nakaya T, de Fabritiis G, Coma M, et al. Amyloid-dependent triosephosphate isomerase nitrotyrosination induces glycation and tau fibrillation. Brain. 2009; 132(Pt 5):1335– 45. Epub 2009/03/03. doi: awp023 [pii] doi: 10.1093/brain/awp023 PMID: 19251756.
- Aguirre Y, Cabrera N, Aguirre B, Perez-Montfort R, Hernandez-Santoyo A, Reyes-Vivas H, et al. Different contribution of conserved amino acids to the global properties of triosephosphate isomerases. Proteins. 2013. Epub 2013/08/24. doi: <u>10.1002/prot.24398</u> PMID: <u>23966267</u>.
- Hrizo SL, Fisher IJ, Long DR, Hutton JA, Liu Z, Palladino MJ. Early mitochondrial dysfunction leads to altered redox chemistry underlying pathogenesis of TPI deficiency. Neurobiol Dis. 2013; 54:289–96. Epub 2013/01/16. doi: S0969-9961(13)00011-9 [pii] doi: <u>10.1016/j.nbd.2012.12.020</u> PMID: <u>23318931</u>; PubMed Central PMCID: PMC3628936.
- Hrizo SL, Palladino MJ. Hsp70- and Hsp90-mediated proteasomal degradation underlies TPI sugarkill pathogenesis in Drosophila. Neurobiol Dis. 2010; 40(3):676–83. Epub 2010/08/24. doi: S0969-9961 (10)00267-6 [pii] doi: <u>10.1016/j.nbd.2010.08.011</u> PMID: <u>20727972</u>; PubMed Central PMCID: PMC2955819.
- **35.** Uversky VN, Fink AL. Protein misfolding, aggregation and conformational diseases. Part B, Molecular mechanisms of conformational diseases. New York; London: Springer; 2007. xxv, 537 p. p.
- 36. Zhang X, Smith DL, Meriin AB, Engemann S, Russel DE, Roark M, et al. A potent small molecule inhibits polyglutamine aggregation in Huntington's disease neurons and suppresses neurodegeneration in vivo. Proc Natl Acad Sci U S A. 2005; 102(3):892–7. Epub 2005/01/12. doi: 0408936102 [pii] doi: 10. 1073/pnas.0408936102 PMID: 15642944; PubMed Central PMCID: PMC545525.
- Dietzl G, Chen D, Schnorrer F, Su KC, Barinova Y, Fellner M, et al. A genome-wide transgenic RNAi library for conditional gene inactivation in Drosophila. Nature. 2007; 448(7150):151–6. Epub 2007/07/ 13. doi: nature05954 [pii] doi: 10.1038/nature05954 PMID: 17625558.

- Komives EA, Chang LC, Lolis E, Tilton RF, Petsko GA, Knowles JR. Electrophilic catalysis in triosephosphate isomerase: the role of histidine-95. Biochemistry. 1991; 30(12):3011–9. Epub 1991/03/26. PMID: 2007138.
- Zhang Z, Sugio S, Komives EA, Liu KD, Knowles JR, Petsko GA, et al. Crystal structure of recombinant chicken triosephosphate isomerase-phosphoglycolohydroxamate complex at 1.8-A resolution. Biochemistry. 1994; 33(10):2830–7. Epub 1994/03/15. PMID: <u>8130195</u>.
- 40. Mande SC, Mainfroid V, Kalk KH, Goraj K, Martial JA, Hol WG. Crystal structure of recombinant human triosephosphate isomerase at 2.8 A resolution. Triosephosphate isomerase-related human genetic disorders and comparison with the trypanosomal enzyme. Protein Sci. 1994; 3(5):810–21. Epub 1994/05/ 01. doi: 10.1002/pro.5560030510 PMID: 8061610; PubMed Central PMCID: PMC2142725.
- **41.** Wierenga RK, Noble ME, Postma JP, Groendijk H, Kalk KH, Hol WG, et al. The crystal structure of the "open" and the "closed" conformation of the flexible loop of trypanosomal triosephosphate isomerase. Proteins. 1991; 10(1):33–49. Epub 1991/01/01. doi: 10.1002/prot.340100105 PMID: 2062827.
- Lodi PJ, Chang LC, Knowles JR, Komives EA. Triosephosphate isomerase requires a positively charged active site: the role of lysine-12. Biochemistry. 1994; 33(10):2809–14. Epub 1994/03/15. PMID: <u>8130193</u>.
- Joseph-McCarthy D, Lolis E, Komives EA, Petsko GA. Crystal structure of the K12M/G15A triosephosphate isomerase double mutant and electrostatic analysis of the active site. Biochemistry. 1994; 33 (10):2815–23. Epub 1994/03/15. PMID: <u>8130194</u>.
- Mainfroid V, Terpstra P, Beauregard M, Frere JM, Mande SC, Hol WG, et al. Three hTIM mutants that provide new insights on why TIM is a dimer. J Mol Biol. 1996; 257(2):441–56. Epub 1996/03/29. doi: S0022-2836(96)90174-6 [pii] doi: 10.1006/jmbi.1996.0174 PMID: 8609635.
- Vijayakrishnan N, Broadie K. Temperature-sensitive paralytic mutants: insights into the synaptic vesicle cycle. Biochem Soc Trans. 2006; 34(Pt 1):81–7. Epub 2006/01/19. doi: BST0340081 [pii] doi: <u>10.1042/</u> BST0340081 PMID: 16417488.
- Brent JR, Werner KM, McCabe BD. Drosophila larval NMJ dissection. J Vis Exp. 2009;(24:). Epub 2009/02/21. doi: 1107 [pii] doi: <u>10.3791/1107</u> PMID: <u>19229190</u>; PubMed Central PMCID: PMC2762896.
- Verstreken P, Ohyama T, Bellen HJ. FM 1–43 labeling of synaptic vesicle pools at the Drosophila neuromuscular junction. Methods Mol Biol. 2008; 440:349–69. Epub 2008/03/29. doi: <u>10.1007/978-1-59745-178-9_26</u> PMID: <u>18369958</u>; PubMed Central PMCID: PMC2712823.
- Saheki Y, De Camilli P. Synaptic vesicle endocytosis. Cold Spring Harb Perspect Biol. 2012; 4(9): a005645. Epub 2012/07/06. doi: cshperspect.a005645 [pii] doi: <u>10.1101/cshperspect.a005645</u> PMID: <u>22763746</u>; PubMed Central PMCID: PMC3428771.
- Wan HI, DiAntonio A, Fetter RD, Bergstrom K, Strauss R, Goodman CS. Highwire regulates synaptic growth in Drosophila. Neuron. 2000; 26(2):313–29. Epub 2000/06/06. doi: S0896-6273(00)81166-6 [pii]. PMID: <u>10839352</u>.
- McCabe BD, Marques G, Haghighi AP, Fetter RD, Crotty ML, Haerry TE, et al. The BMP homolog Gbb provides a retrograde signal that regulates synaptic growth at the Drosophila neuromuscular junction. Neuron. 2003; 39(2):241–54. Epub 2003/07/23. doi: S0896627303004264 [pii]. PMID: <u>12873382</u>.
- Packard M, Koo ES, Gorczyca M, Sharpe J, Cumberledge S, Budnik V. The Drosophila Wnt, wingless, provides an essential signal for pre- and postsynaptic differentiation. Cell. 2002; 111(3):319–30. Epub 2002/11/07. doi: S0092867402010474 [pii]. PMID: 12419243; PubMed Central PMCID: PMC3499980.
- Borchert TV, Zeelen JP, Schliebs W, Callens M, Minke W, Jaenicke R, et al. An interface point-mutation variant of triosephosphate isomerase is compactly folded and monomeric at low protein concentrations. FEBS Lett. 1995; 367(3):315–8. Epub 1995/07/03. doi: 0014-5793(95)00586-X [pii]. PMID: <u>7607330</u>.
- Mainfroid V, Mande SC, Hol WG, Martial JA, Goraj K. Stabilization of human triosephosphate isomerase by improvement of the stability of individual alpha-helices in dimeric as well as monomeric forms of the protein. Biochemistry. 1996; 35(13):4110–7. Epub 1996/04/02. doi: <u>10.1021/bi952692n</u> bi952692n [pii]. PMID: 8672446.
- Hinault MP, Ben-Zvi A, Goloubinoff P. Chaperones and proteases: cellular fold-controlling factors of proteins in neurodegenerative diseases and aging. J Mol Neurosci. 2006; 30(3):249–65. Epub 2007/ 04/03. doi: JMN:30:3:249 [pii] doi: <u>10.1385/JMN:30:3:249</u> PMID: <u>17401151</u>.
- Lambeir AM, Backmann J, Ruiz-Sanz J, Filimonov V, Nielsen JE, Kursula I, et al. The ionization of a buried glutamic acid is thermodynamically linked to the stability of Leishmania mexicana triose phosphate isomerase. Eur J Biochem. 2000; 267(9):2516–24. Epub 2000/04/28. doi: ejb1254 [pii]. PMID: 10785370.
- 56. Eber SW, Pekrun A, Bardosi A, Gahr M, Krietsch WK, Kruger J, et al. Triosephosphate isomerase deficiency: haemolytic anaemia, myopathy with altered mitochondria and mental retardation due to a new

variant with accelerated enzyme catabolism and diminished specific activity. Eur J Pediatr. 1991; 150 (11):761–6. Epub 1991/09/01. PMID: <u>1959537</u>.

- Chang ML, Artymiuk PJ, Wu X, Hollan S, Lammi A, Maquat LE. Human triosephosphate isomerase deficiency resulting from mutation of Phe-240. Am J Hum Genet. 1993; 52(6):1260–9. Epub 1993/06/ 01. PMID: 8503454; PubMed Central PMCID: PMC1682273.
- Pretsch W. Triosephosphate isomerase activity-deficient mice show haemolytic anaemia in homozygous condition. Genet Res (Camb). 2009; 91(1):1–4. Epub 2009/02/18. doi: S0016672308009944 [pii] doi: <u>10.1017/S0016672308009944</u> PMID: <u>19220926</u>.
- Ahmed N, Battah S, Karachalias N, Babaei-Jadidi R, Horanyi M, Baroti K, et al. Increased formation of methylglyoxal and protein glycation, oxidation and nitrosation in triosephosphate isomerase deficiency. Biochim Biophys Acta. 2003; 1639(2):121–32. Epub 2003/10/16. doi: S0925443903001406 [pii]. PMID: 14559119.
- Gruning NM, Du D, Keller MA, Luisi BF, Ralser M. Inhibition of triosephosphate isomerase by phosphoenolpyruvate in the feedback-regulation of glycolysis. Open Biol. 2014; 4:130232. Epub 2014/03/07. doi: rsob.130232 [pii] doi: <u>10.1098/rsob.130232</u> PMID: <u>24598263</u>; PubMed Central PMCID: PMC3971408.
- Gruning NM, Rinnerthaler M, Bluemlein K, Mulleder M, Wamelink MM, Lehrach H, et al. Pyruvate kinase triggers a metabolic feedback loop that controls redox metabolism in respiring cells. Cell Metab. 2011; 14(3):415–27. Epub 2011/09/13. doi: S1550-4131(11)00300-7 [pii] doi: <u>10.1016/j.cmet.2011.06.</u> <u>017</u> PMID: <u>21907146</u>; PubMed Central PMCID: PMC3202625.
- Trushina E, McMurray CT. Oxidative stress and mitochondrial dysfunction in neurodegenerative diseases. Neuroscience. 2007; 145(4):1233–48. Epub 2007/02/17. doi: S0306-4522(06)01433-3 [pii] doi: 10.1016/j.neuroscience.2006.10.056 PMID: 17303344.
- Lin MT, Beal MF. Mitochondrial dysfunction and oxidative stress in neurodegenerative diseases. Nature. 2006; 443(7113):787–95. Epub 2006/10/20. doi: nature05292 [pii] doi: <u>10.1038/nature05292</u> PMID: <u>17051205</u>.
- Kim GH, Kim JE, Rhie SJ, Yoon S. The Role of Oxidative Stress in Neurodegenerative Diseases. Exp Neurobiol. 2015; 24(4):325–40. Epub 2015/12/30. doi: <u>10.5607/en.2015.24.4.325</u> PMID: <u>26713080</u>; PubMed Central PMCID: PMC4688332.
- Smith CA, Kortemme T. Backrub-like backbone simulation recapitulates natural protein conformational variability and improves mutant side-chain prediction. J Mol Biol. 2008; 380(4):742–56. Epub 2008/06/ 13. doi: S0022-2836(08)00577-9 [pii] doi: <u>10.1016/j.jmb.2008.05.023</u> PMID: <u>18547585</u>; PubMed Central PMCID: PMC2603262.
- Suzuki DT, Grigliatti T, Williamson R. Temperature-sensitive mutations in Drosophila melanogaster. VII. A mutation (para-ts) causing reversible PMID: <u>adult</u> paralysis. Proc Natl Acad Sci U S A. 1971; 68 (5):890–3. Epub 1971/05/01. PMID: <u>5280526</u>; PubMed Central PMCID: PMC389073.
- Jackson FR, Wilson SD, Strichartz GR, Hall LM. Two types of mutants affecting voltage-sensitive sodium channels in Drosophila melanogaster. Nature. 1984; 308(5955):189–91. Epub 1984/03/08. PMID: <u>6322008</u>.
- Kawasaki F, Felling R, Ordway RW. A temperature-sensitive paralytic mutant defines a primary synaptic calcium channel in Drosophila. J Neurosci. 2000; 20(13):4885–9. Epub 2000/06/24. doi: 20/13/4885 [pii]. PMID: 10864946.
- Palladino MJ, Bower JE, Kreber R, Ganetzky B. Neural dysfunction and neurodegeneration in Drosophila Na+/K+ ATPase alpha subunit mutants. J Neurosci. 2003; 23(4):1276–86. Epub 2003/02/25. doi: 23/4/1276 [pii]. PMID: <u>12598616</u>.
- Pallanck L, Ordway RW, Ganetzky B. A Drosophila NSF mutant. Nature. 1995; 376(6535):25. Epub 1995/07/06. doi: <u>10.1038/376025a0</u> PMID: <u>7596428</u>.
- Siddiqi O, Benzer S. Neurophysiological defects in temperature-sensitive paralytic mutants of Drosophila melanogaster. Proc Natl Acad Sci U S A. 1976; 73(9):3253–7. Epub 1976/09/01. PMID: <u>184469</u>; PubMed Central PMCID: PMC430996.
- 72. Scott DA, Das U, Tang Y, Roy S. Mechanistic logic underlying the axonal transport of cytosolic proteins. Neuron. 2011; 70(3):441–54. Epub 2011/05/11. doi: S0896-6273(11)00295-9 [pii] doi: <u>10.1016/j.</u> <u>neuron.2011.03.022</u> PMID: <u>21555071</u>; PubMed Central PMCID: PMC3096075.
- Boulanger A, Farge M, Ramanoudjame C, Wharton K, Dura JM. Drosophila motor neuron retraction during metamorphosis is mediated by inputs from TGF-beta/BMP signaling and orphan nuclear receptors. PLoS One. 2012; 7(7):e40255. Epub 2012/07/14. doi: <u>10.1371/journal.pone.0040255</u> PONE-D-12-01465 [pii]. PMID: <u>22792255</u>; PubMed Central PMCID: PMC3390346.
- Miller D, Hannon C, Ganetzky B. A mutation in Drosophila Aldolase causes temperature-sensitive paralysis, shortened lifespan, and neurodegeneration. J Neurogenet. 2012; 26(3–4):317–27. Epub

2012/08/14. doi: 10.3109/01677063.2012.706346 PMID: 22882183; PubMed Central PMCID: PMC3597222.

- 75. Wang P, Saraswati S, Guan Z, Watkins CJ, Wurtman RJ, Littleton JT. A Drosophila temperature-sensitive seizure mutant in phosphoglycerate kinase disrupts ATP generation and alters synaptic function. J Neurosci. 2004; 24(19):4518–29. Epub 2004/05/14. doi: <u>10.1523/JNEUROSCI.0542-04.2004</u> 24/19/ 4518 [pii]. PMID: <u>15140922</u>.
- 76. Rangaraju V, Calloway N, Ryan TA. Activity-driven local ATP synthesis is required for synaptic function. Cell. 2014; 156(4):825–35. Epub 2014/02/18. doi: S0092-8674(14)00013-0 [pii] doi: <u>10.1016/j.cell.</u> <u>2013.12.042</u> PMID: <u>24529383</u>.
- Chavan V, Willis J, Walker SK, Clark HR, Liu X, Fox MA, et al. Central presynaptic terminals are enriched in ATP but the majority lack mitochondria. PLoS One. 2015; 10(4):e0125185. Epub 2015/05/ 01. doi: 10.1371/journal.pone.0125185 PONE-D-14-47989 [pii]. PMID: 25928229; PubMed Central PMCID: PMC4416033.
- Verstreken P, Ly CV, Venken KJ, Koh TW, Zhou Y, Bellen HJ. Synaptic mitochondria are critical for mobilization of reserve pool vesicles at Drosophila neuromuscular junctions. Neuron. 2005; 47(3):365– 78. Epub 2005/08/02. doi: S0896-6273(05)00523-4 [pii] doi: <u>10.1016/j.neuron.2005.06.018</u> PMID: <u>16055061</u>.
- Hall CN, Klein-Flugge MC, Howarth C, Attwell D. Oxidative phosphorylation, not glycolysis, powers presynaptic and postsynaptic mechanisms underlying brain information processing. J Neurosci. 2012; 32 (26):8940–51. Epub 2012/06/30. doi: 32/26/8940 [pii] doi: <u>10.1523/JNEUROSCI.0026-12.2012</u> PMID: 22745494; PubMed Central PMCID: PMC3390246.
- Pathak D, Shields L, Mendelsohn BA, Haddad D, Lin W, Gerencser AA, et al. The Role of Mitochondrially Derived ATP in Synaptic Vesicle Recycling. J Biol Chem. 2015. Epub 2015/07/02. doi: M115.656405 [pii] doi: 10.1074/jbc.M115.656405 PMID: 26126824.
- Harris JJ, Jolivet R, Attwell D. Synaptic energy use and supply. Neuron. 2012; 75(5):762–77. Epub 2012/09/11. doi: S0896-6273(12)00756-8 [pii] doi: <u>10.1016/j.neuron.2012.08.019</u> PMID: <u>22958818</u>.
- Schurr A, West CA, Rigor BM. Lactate-supported synaptic function in the rat hippocampal slice preparation. Science. 1988; 240(4857):1326–8. Epub 1988/06/03. PMID: <u>3375817</u>.
- Wyss MT, Jolivet R, Buck A, Magistretti PJ, Weber B. In vivo evidence for lactate as a neuronal energy source. J Neurosci. 2011; 31(20):7477–85. Epub 2011/05/20. doi: 31/20/7477 [pii] doi: <u>10.1523/</u> JNEUROSCI.0415-11.2011 PMID: <u>21593331</u>.
- Ng J, Luo L. Rho GTPases regulate axon growth through convergent and divergent signaling pathways. Neuron. 2004; 44(5):779–93. Epub 2004/12/02. doi: S0896627304007524 [pii] doi: <u>10.1016/j.neuron.</u> 2004.11.014 PMID: 15572110.
- Wang D, Zhang L, Zhao G, Wahlstrom G, Heino TI, Chen J, et al. Drosophila twinfilin is required for cell migration and synaptic endocytosis. J Cell Sci. 2010; 123(Pt 9):1546–56. Epub 2010/04/23. doi: 123/9/ 1546 [pii] doi: <u>10.1242/jcs.060251</u> PMID: <u>20410372</u>.
- Abe T, Yamazaki D, Murakami S, Hiroi M, Nitta Y, Maeyama Y, et al. The NAV2 homolog Sickie regulates F-actin-mediated axonal growth in Drosophila mushroom body neurons via the non-canonical Rac-Cofilin pathway. Development. 2014; 141(24):4716–28. Epub 2014/11/21. doi: dev.113308 [pii] doi: 10.1242/dev.113308 PMID: 25411210.
- Gorlich A, Wolf M, Zimmermann AM, Gurniak CB, Al Banchaabouchi M, Sassoe-Pognetto M, et al. N-cofilin can compensate for the loss of ADF in excitatory synapses. PLoS One. 2011; 6(10):e26789. Epub 2011/11/03. doi: <u>10.1371/journal.pone.0026789</u> PONE-D-11-13320 [pii]. PMID: <u>22046357</u>; PubMed Central PMCID: PMC3203908.
- Wolf M, Zimmermann AM, Gorlich A, Gurniak CB, Sassoe-Pognetto M, Friauf E, et al. ADF/Cofilin Controls Synaptic Actin Dynamics and Regulates Synaptic Vesicle Mobilization and Exocytosis. Cereb Cortex. 2015; 25(9):2863–75. Epub 2014/04/29. doi: bhu081 [pii] doi: <u>10.1093/cercor/bhu081</u> PMID: <u>24770705</u>.
- Jung J, Yoon T, Choi EC, Lee K. Interaction of cofilin with triose-phosphate isomerase contributes glycolytic fuel for Na,K-ATPase via Rho-mediated signaling pathway. J Biol Chem. 2002; 277(50):48931– 7. Epub 2002/10/03. doi: <u>10.1074/jbc.M208806200</u> M208806200 [pii]. PMID: <u>12359716</u>.
- Huang J, Zhou W, Dong W, Watson AM, Hong Y. From the Cover: Directed, efficient, and versatile modifications of the Drosophila genome by genomic engineering. Proc Natl Acad Sci U S A. 2009; 106 (20):8284–9. Epub 2009/05/12. doi: 0900641106 [pii] doi: <u>10.1073/pnas.0900641106</u> PMID: <u>19429710</u>; PubMed Central PMCID: PMC2688891.
- Huang J, Zhou W, Watson AM, Jan YN, Hong Y. Efficient ends-out gene targeting in Drosophila. Genetics. 2008; 180(1):703–7. Epub 2008/09/02. doi: genetics.108.090563 [pii] doi: <u>10.1534/genetics.108.</u> <u>090563</u> PMID: <u>18757917</u>; PubMed Central PMCID: PMC2535722.

- 92. Williams JC, Zeelen JP, Neubauer G, Vriend G, Backmann J, Michels PA, et al. Structural and mutagenesis studies of leishmania triosephosphate isomerase: a point mutation can convert a mesophilic enzyme into a superstable enzyme without losing catalytic power. Protein Eng. 1999; 12(3):243–50. Epub 1999/05/11. PMID: 10235625.
- 93. Ganetzky B, Wu CF. Indirect Suppression Involving Behavioral Mutants with Altered Nerve Excitability in DROSOPHILA MELANOGASTER. Genetics. 1982; 100(4):597–614. Epub 1982/04/01. PMID: <u>17246073</u>; PubMed Central PMCID: PMC1201835.
- 94. Cromey DW. Avoiding twisted pixels: ethical guidelines for the appropriate use and manipulation of scientific digital images. Sci Eng Ethics. 2010; 16(4):639–67. Epub 2010/06/23. doi: <u>10.1007/s11948-010-9201-y</u> PMID: <u>20567932</u>; PubMed Central PMCID: PMC4114110.
- Otwinowski Z, Minor W. Processing of X-ray diffraction data collected in oscillation mode. Method Enzymol. 1997; 276:307–26. doi: 10.1016/S0076-6879(97)76066-X PMID: <u>ISI:A1997BH42P00020</u>.
- 96. Adams PD, Afonine PV, Bunkoczi G, Chen VB, Davis IW, Echols N, et al. PHENIX: a comprehensive Python-based system for macromolecular structure solution. Acta Crystallogr D Biol Crystallogr. 2010; 66(Pt 2):213–21. Epub 2010/02/04. doi: S0907444909052925 [pii] doi: <u>10.1107/S0907444909052925</u> PMID: <u>20124702</u>; PubMed Central PMCID: PMC2815670.
- Davis IW, Leaver-Fay A, Chen VB, Block JN, Kapral GJ, Wang X, et al. MolProbity: all-atom contacts and structure validation for proteins and nucleic acids. Nucleic Acids Res. 2007; 35(Web Server issue): W375–83. Epub 2007/04/25. doi: gkm216 [pii] doi: <u>10.1093/nar/gkm216</u> PMID: <u>17452350</u>; PubMed Central PMCID: PMC1933162.
- Lauck F, Smith CA, Friedland GF, Humphris EL, Kortemme T. RosettaBackrub—a web server for flexible backbone protein structure modeling and design. Nucleic Acids Res. 2010; 38(Web Server issue): W569–75. Epub 2010/05/14. doi: gkq369 [pii] doi: <u>10.1093/nar/gkq369</u> PMID: <u>20462859</u>; PubMed Central PMCID: PMC2896185.